

Contents lists available at ScienceDirect

# Technology in Society



journal homepage: www.elsevier.com/locate/techsoc

# Internet over the ocean: A smart IoT-enabled digital ecosystem for empowering coastal fisher communities

Sruthy Anand<sup>a</sup>, Mojtaba Enayati<sup>b</sup>, Dhanesh Raj<sup>a</sup>, Alberto Montresor<sup>c</sup>, Maneesha Vinodini Ramesh<sup>a,b,\*</sup>

<sup>a</sup> Center for Wireless Networks & Applications (WNA), Amrita Vishwa Vidyapeetham, Amritapuri, India

<sup>b</sup> Amrita School for Sustainable Futures, Amrita Vishwa Vidyapeetham, Amritapuri, India

<sup>c</sup> Department of Information Engineering and Computer Science, University of Trento Trento, Italy

#### ARTICLE INFO

Keywords: Sustainable social change Coastal community empowerment Digital ecosystem Blockchain technology Co-Design Digital divide Fishing Community

# ABSTRACT

Coastal communities across the globe are prone to imminent disasters and emergencies due to their geographic placement, and their fishing occupations add to their exposure to ocean-related disasters more during their occupational voyage. The fishermen in the state of Kerala, India, lead a distinctive lifestyle compared to the mainstream community, primarily because of a significant digital divide. This divide hampers their access to livelihood support, such as live market information and impedes any potential improvement in their quality of life through digital services. The community's transformation is possible only through discrete and gradual steps of progression by bringing services to different areas such as education, connectivity, technology adoption, business, and livelihood. This study revealed prevalent gaps in the fishing community, including a lack of digital access and communication opportunities in the deep ocean, a deficiency in digital capacity preventing them from leveraging the benefits of Information and communications technology (ICT), leading to community isolation from digital inclusion and limiting economic opportunities. To address these issues, the study proposes an IoTenabled digital ecosystem, transforming the community into a smart digital community. This ecosystem integrates a low-cost deep ocean communication solution, supportive software applications, a blockchain-based ecommerce platform, and a sustainable social change model for technology adoption. Proper adoption and utilisation of the digital ecosystem can strengthen the community's resilience to natural disasters, and climatic changes by providing relevant information and livelihood resilience by offering them an e-commerce platform to improve income and reduce their vulnerability. It will equip them to enhance their responsiveness to changing environmental and social factors and bounce back to build their innate resilience.

#### 1. Introduction

With a coastline that stretches over 7500 km and the third-highest fish production in the world, India is home to more than 28 million fishermen, the majority of whom are small-scale fishermen [1,2]. According to the Indian Ministry of Fisheries [2] there are 3461 marine fishing villages with 1457 marine fish landing sites spanning nine maritime states and two union territories in India, the majority of which belong to the rural community [3]. Fishing is the key source of employment for millions of coastal people, approximately 4 million fishers, and 85 % of them have fishing as their primary source of income [4] thus contributing significantly to the country's food security and economy. Generally, fishing is considered one of the world's most unreliable, unstable, and unsafe professions [5]. The dynamic variability of the ocean weather such as marine heat waves, and sea swell, impacts the spatiotemporal distribution of fish stocks [6]. It also increases the occurrence of emergencies such as the capsizing of fishing vessels, making this livelihood an unsafe occupation. The fishing community experiences a remarkably high loss of lives, with approximately 100,000 fishing-related deaths occurring globally each year [7], making it a particularly perilous occupation. Generally, for medium-sized fishing vessels (less than 100 feet in length), the fishermen can be away from the shore for 7–15 days and travel 120 km on average while at sea. During this time, they face unpredictable weather and extreme environmental conditions, without having a reliable communication facility. This lack

https://doi.org/10.1016/j.techsoc.2024.102686

Received 2 February 2023; Received in revised form 13 July 2024; Accepted 14 August 2024 Available online 20 August 2024

<sup>\*</sup> Corresponding author. E-mail address: maneesha@amrita.edu (M.V. Ramesh).

<sup>0160-791</sup>X/© 2024 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC license (http://creativecommons.org/licenses/by-nc/4.0/).

of communication makes them unable to seek help during emergencies. Additionally, they are exposed to life-threatening dangers, including unpredictable collisions with other fishing vessels or ships (if their vessel is in the ship routes), engine failures, and unexpected situations such as pirate attacks and unintentional maritime border crossings, potentially leading to captivation by the neighbouring border force [8]. This unreliability, safety issues, and lack of reliable communication add up to their occupational vulnerability.

A livelihood is considered resilient if the community or individual practising it has proper access to human, physical, social, natural, and financial resources to build the capacity to recover from various shocks and disturbances, such as natural disasters, economic downturns, and conflicts with other stakeholders [9]. Additionally, according to the Sustainable Livelihoods Analytical Framework by the Department for International Development (DFID), the livelihood resilience of a household can be improved through income improvement, well-being improvement, reduced vulnerability, and resource optimisation [10]. In the context of the fishery as a livelihood, the current situation in the study community highlights their lack of access to digital services to build resilience capacity. These services could assist them in coping with unfavourable weather conditions at sea, such as accessing related weather forecasting information for the prevention of potential health hazards, addressing their over-dependence on middlemen who control the fish market, serving as the most influential stakeholders to reduce their vulnerability and increase their profit margin, and monitoring the marine resource depletion for optimised resource management policies [11].

One of the primary obstacles to enhancing fishermen's livelihood resilience, safety, and digital empowerment is the absence of digital services specifically designed for them. These services would aid them in various instances, such as accessing a wider range of customers without the interference of middlemen, providing live market information for informed decision-making, offering emergency information in the ocean, and providing Internet access deep in the ocean. Beyond 15 km from the shore, cellular coverage and Internet connectivity are unavailable. This makes them completely isolated from their community during the trip, lacking real-time business management from the ocean. The lack of real-time communication opportunities forces fishermen to rely on intermediaries at the fishing harbours, who provide periodic messages and advice based on their connectivity and market analysis. However, these services come at a cost, and often, intermediaries take a share of the fisherman's profit. This exploitation occurs due to the fishermen's limited access to real-time information during fishing trips.

Enhancements in weather forecasting systems and ocean information services have been devised to provide early warnings of potential disasters and emergencies in the ocean. However, currently, during emergencies such as high tides, collisions between two fishing vessels, ship-to-boat collisions, bad weather, or cyclone formation, timely information cannot be effectively disseminated to the stakeholders at the shore due to the unavailability of the Internet over the ocean beyond 15 km from the shore. Similarly, stakeholders onshore, such as administrators from the Department of Fisheries, fishing vessel owners, and family members of the crewmen, currently lack a solution that could initiate real-time tracking and monitoring for fishing vessels. However, there are some existing services such as mKRISHI Fisheries App [12], Nautide [13], Sagara, Search and Rescue Aid Tool (SARAT) [14], INCOIS feedback app, Tuna (PFZ) services, "Sagar Vani" - an integrated information dissemination system [15], Windy.app [16], Fisher Friend Mobile Application (FFMA) [17], etc., that can be used where the cellular network coverage is available. Even so, the fishermen are hesitant or unable to use them due to the lack of digital access, knowledge, and skills to use digital technologies and services.

Lack of digital literacy makes the fishermen rely solely on conventional methods for their livelihood and business, impacting their resilience, sustainability, and overall quality of life. The conventional methods, which rely on the expertise of fishermen to anticipate sea state and weather conditions, determine the spatial location of fish, and gather information from auction dealers, are commonly used for establishing dynamic pricing when selling the daily catch and deciding when to return to the shore. However, these conventional methods cannot provide real-time information on changes in weather, sea state, market dynamics, etc., thereby affecting the resilience of the fishermen's livelihood system. The limitations of these conventional methods make fishermen susceptible to exploitation by intermediaries and restrict their access to a broader consumer base, unlike business owners using ecommerce platforms.

Digital literacy empowers fishermen to diversify their livelihood opportunities, leading to increased profit margins. For instance, the increased use of digital payments in rural areas has led to the growth of the digital economy, creating new opportunities for businesses and entrepreneurs and improving livelihoods [18]. A study by Ye and Yang (2020) [19] presents a revelatory case whereby a mobile platform empowers rural people through structural, psychological and resource empowerment. Several smartphone applications are designed to empower the rural population in various fields such as delivering timely information, providing market information and healthcare services [20]. This, in turn, improves their overall quality of life by allowing them to achieve greater profits with the same effort or maintain the same profit with less effort. These observations also highlight the digital divide within the community, stemming from the lack of digital access at sea and insufficient digital skills to utilise available technologies for enhanced safety and resilience. In the literature, the digital divide is defined as the inequalities that exist in access to and use of the Internet [21] and other ICT opportunities. The current digital divide exacerbates the communities' ability to be inclusive of the mainstream community. Specifically, it hampers the ability of fishermen communities to integrate with urban and suburban communities that have 24x7 digital access. This significantly impacts the digital outcomes of the community. Here, digital outcomes refer to using ICT to access information online to enhance their livelihood and safety. These challenges can be addressed through the development of a digital ecosystem tailored for the fisher community.

This research focuses on addressing the aforementioned challenges, including the lack of connectivity at sea (i.e., Digital Access), limited skills to use digital services (i.e., Digital Capacity, which is the ability to use and perceive the possibilities of the technology) and lack of real-time market information for dynamic decision making by introducing the "OceanNet Digital Ecosystem (ODE)." The ODE comprises multiple solutions designed to meet the specific needs of fishermen, ultimately providing a pathway to have a communication medium at sea for enhancing their safety, and livelihood resilience through digital advancements. A digital ecosystem is a modular, digitally connected, loose network of interacting organizations or a collection of platforms, software, and applications specific to it [22,23]; therefore, the proposed solution is called a digital ecosystem.

The research team has developed a real-time communication system for fishermen that provides digital connectivity and helps them receive emergency warning services. The first part of the solution is an IoTbased architecture to provide Internet access over the ocean. It has been designed and developed to enhance digital access. Initially, the system was tested by researchers through deployment on three fishing vessels, conducting sea trials to verify its envisioned long-range connectivity. Subsequently, the system was further deployed on eleven additional fishing vessels, with a total of fourteen vessels utilising it during their regular fishing trips. The research has also produced a realtime tracking mobile application aimed at enhancing safety at sea. The IoT system, facilitating real-time communication, is integrated with web and mobile platforms, effectively improving digital access opportunities. Engaging with multiple stakeholders has boosted their motivation to use the system, and regular use has honed their skills in leveraging it for communication, safety, and overall resilience.

The digital outcomes of the OceanNet Digital Ecosystem (ODE),

specifically increased safety at sea and enhanced livelihood resilience, are further fortified through the integration of blockchain technology into its design. Blockchain technology serves as the backbone for establishing a middleman-free online e-commerce framework, ultimately enhancing fishermen's livelihood opportunities. This, in turn, contributes to bolstering their overall resilience against coastal risks and reducing dependence on intermediaries, thereby impacting their economic stability positively. Utilising a Blockchain-based e-commerce platform for fish sales can significantly enhance the resilience of fishermen's livelihoods by addressing several key factors. Firstly, removing intermediaries eliminates exploitation and allows them to capture a larger share of the profits, increasing their financial capital and reducing vulnerability to income shocks [24]. Secondly, reaching a wider customer base through the platform expands their market access and reduces dependence on local fluctuations, improving their ability to adapt to changing market conditions [11,25]. Thirdly, clear price determination based on market demand and automated auctions empowers fishermen with greater control over pricing (resistance), fostering a more equitable system [9]. This can also potentially increase income, reduce their vulnerability, and improve their capacity to absorb shocks [26]. Overall, the platform offers a diversification strategy [27] by providing a new income stream, strengthening their overall livelihood resilience.

This comprehensive digital ecosystem architecture integrates opportunities to enhance digital access, digital capacity, and digital outcomes. The proposed ecosystem empowers fishermen and other stakeholders to access digital services and analyse spatiotemporal patterns of the fishing vessels, which can later be utilised to plan future fishing trips. It also enables them to understand the interrelationships between different factors, such as fish availability, type of fish, and market demands, which can be considered to plan future activities, ultimately enhancing the livelihood of the fisherman community. Livelihood involves specific capabilities, activities, and assets that are required to earn for living [28]. The proposed ODE focuses on both safety and livelihood. However, the success of the proposed ecosystem is dependent on the technology adoption by the fishing community. This has been considered from the initiation of the research work. The realisation of the envisaged digital outcome hinges upon the active utilisation of the digital ecosystem by fishermen, their families, vessel owners, and other stakeholders. This usage is vital for communication, livelihood, safety, and overall improvement of resilience and quality of life within the community. The adoption of new technology necessitates a model that incorporates social elements, including local actors, the environment, culture, traditional behaviours, and motivational factors [29].

The above-considered points have specifically been integrated into this initiative, and this paper explains the approaches used in detail. To achieve this, an optimal technological platform was constructed based on community-centred methods for an assured alignment with their real-life challenges. A sustainable social change model, inspired by the Technology Adoption Model (TAM) [30] and based on community characteristics, abilities, and motivation to adopt new technology, has been proposed. The sustainable social change model envisions developing a set of associated methodologies and a roadmap to achieve the envisioned digital empowerment through community participation and engagement with the technological services through the access provided by the network. The researchers assume that the desired digital outcome will be achieved through the successful prolonged usage of the technology and social pressure to use it. This study details the efforts to develop a smart fishing community, equipping them to use digital resources and services during their occupational voyage to receive timely emergency warning messages and maintain communication with the shore. Furthermore, this paper presents the lessons learned from the problems identified, the steps taken to improve the solution, the technology diffusion process, and the sustainable social change model developed. These elements collectively equip the community towards

becoming a sustainable smart coastal fishing community.

Based on the existing challenges discussed above and the research gaps identified and presented in section 3, the key research questions considered for reducing the digital divide and empowering the fisher community are as follows.

- 1. How can a digital ecosystem be developed to achieve real-time disaster risk reduction and enhance resilience to ocean emergencies?
- 2. How can the fishing community be empowered to utilise the developed technologies for enhanced safety and livelihood?

The rest of the paper is organised as follows: Section 2 details the methods used during the different phases of this research. Section 3 reviews the current related literature and presents the identified gaps. Section 4 presents the research results and provides a detailed description of the architecture and components of the digital ecosystem and its implementation details. Section 5 provides a discussion on how the research questions have been answered in the study. Section 6 presents the limitations of the study and Section 7 concludes the paper with a summary of its findings and presents the potential future directions to enhance the digital ecosystem for coastal fishing communities.

# 2. Methods

To tackle the research questions outlined in Section 1, the study employed three main categories of methods: desk literature review, codesign process, and a sustainable social change model. Fig. 1 provides an overview of these methods, presented in the order they were utilised in the study.

Below, a detailed overview of the research community and the methodologies used in this study is provided.

# 2.1. Study area

As illustrated in Fig. 2, the selected study area for the proposed ODE implementation is the coastal village of Alappad in the State of Kerala, India, with village coordinates at 09.0898° N, 76.4864° E. This village is a narrow stretch, approximately 17 km long, extending from south to north. According to the last census of India [31], the village is home to 21,655 residents, with an almost equal distribution between males and females. The community members in this study share a common set of socio-cultural standards and beliefs, forming a close-knit community characterized by shared socio-cultural values. Given that the majority of the male residents in the village are fishermen, this study area catalyzes nearby coastal communities to embrace new digital practices. These practices aim to support their livelihoods, enhance livelihood resilience, and provide disaster alerts and offshore emergency communication facilities. This focused study on Alappad offers insights that can be extrapolated to benefit neighbouring coastal communities, illustrating the potential for widespread adoption of digital solutions in support of sustainable livelihoods and improved resilience.

#### 2.2. Literature review

The literature review for this study covered a broad spectrum of sources, comprising the UN and government reports, scholarly articles about ocean communication technologies, blockchain technology, technology adoption models, social change theories, sustainable development, the digital divide, participatory methods (e.g., Participatory Rural Appraisal tools), and co-design approaches. These sources were drawn from digital databases, including government official websites, Scopus, Google Scholar, United Nations reports, and private sector resources. The search results from these digital databases were evaluated within the context of the study's research community, and irrelevant findings were filtered out before further analysis of the articles.



Fig. 1. Overview of the methods utilised in the study.



Fig. 2. Map of the study area.

#### 2.3. Co-design process

The research employed a co-design [32,33] methodology to understand the community's characteristics, chal-lenges, and requirements, enabling the formulation of solutions for the identified issues. Fig. 3 illustrates the co-design process used in developing the OceanNet Digital Ecosystem (ODE) tailored for the fishing community. This method involves active engagement with community stakeholders throughout the design and development phases, ensuring alignment with their requirements and values. Such an approach facilitates a smoother technology adoption process. Extensive engagement with community members occurred during the requirement analysis and problem identification stages, providing a comprehensive understanding of their challenges and digital gaps. The gathered information served as the



Fig. 3. Co-design process for a sustainable solution.

foundation for designing solutions to address the identified issues. Continuous feedback from the community played a pivotal role in refining the proposed solution, contributing to its ongoing improvement. This iterative approach significantly strengthened the capacity to design multiple sustainable solutions, thereby achieving the envisioned digital outcomes. The co-design methodology consists of six phases: issue identification, data analysis, solution development, product validation, solution deployment, feedback collection, and solution refinement.

#### 2.3.1. Data collection

Taking into account the co-design process depicted in Fig. 3, the initial focus of the study was on gathering information from the community and identifying stakeholders for active participation in subsequent stages. This phase was enriched by qualitative, structured, and semi-structured face-to-face interviews, which served as key tools for community engagement. Initiated in 2014, the study has continued to evolve, with ongoing refinements to the solution.

Face-to-face interviews were conducted with over 200 participants, comprising boat owners, captains, crewmen, fishermen, fishermen's family members, and fishing vessel managers. Participants were provided with a detailed explanation of the study's purpose, and their consent was obtained for both interviews and the use of acquired materials. Employing a convenience sampling method, the research team obtained contact numbers from participants through their colleagues within the community. Interviews were conducted in the local language (Malayalam), with translators present for team members unfamiliar with the language. Semi-structured interviews with fishermen participants were conducted at their fishing vessels to gain deeper insights into their practices and daily routines. The interviews lasted a maximum of 30 min for family members and 50 min for other participants. Questions covered various aspects of the fishing process, including the technologies in use, the number of fishermen on each vessel, crewmen's roles and responsibilities, and communication methods used during voyages, both with their families and in emergencies. Additionally, business-related topics, such as fish catch procedures, conventional selling practices, influential stakeholders in the market, and methods to reach customers, were also explored.

This comprehensive process facilitated meticulous data collection, necessary for a profound understanding of the dynamics and needs of the fishing community. Interviews were documented in both audio and written forms, supplemented by captured images of existing technologies on fishing vessels and the stakeholders for further study. The amassed information was then leveraged to design a second set of interviews. The next round of interviews employed a structured qualitative approach based on various indicators related to their occupational, emergency, and personal requirements, as illustrated in Table 1. When conducting interviews with the fishermen's family members, the questionnaire aimed to capture the perspectives of the females within the community. Research volunteers obtained contact information for the fishermen's families through the local Panchayat member (the local selfgovernance structure) and conducted the interviews based on his recommendations.

#### 2.3.2. Data analysis

The interviews were concluded upon reaching a point of repetitive responses from participants. Subsequently, the interviews were transcribed verbatim and stored as documents for further analysis. The information gleaned from these interviews underwent analysis using a manual technique [34,35]. The research team meticulously reviewed the data multiple times, focusing on addressing the research questions and gaining insights into the community's characteristics, capabilities, challenges, and motivations. The outcome of this phase was an understanding of (a) the community's needs, (b) existing communication methodologies prevalent in the community, (c) emergency scenarios,(d) fishing practices, (e) fishing trip patterns, (f) livelihood opportunities,

#### Table 1

The c	uestionnaire	set for	collecting	information	from	the study	v community.
inc c	aconomiune	000101	concernis	momunon	mom	uic otuu	community.

-		0 3 3
SI No	Question indicators	Interview questions
1	E01	Would you like to receive alerts and notifications for weather changes?
2	E02	Do you want to receive alerts on cyclones and tsunamis in advance?
3	E03	What are the early warning systems you currently use at sea?
4	E04	Would you like to receive real-time updates in case of a family emergency? Would you like to be informed about a potential collision with another ship while on a shipping route as a fishing vessel?
5	E05	Would you like to be informed about a potential collision with another ship while on a shipping route as a fishing vessel?
6	E06	Have you experienced any health emergencies at sea during fishing?
7	001	Would you like to be informed about nearby fishing vessels to assist you in the case of food or fuel shortage on your fishing yessel?
8	002	How is fishing done during a single trip? Is the fishing vessel anchored in one place, or does it move? If it moves, what is the movement actuary foregoing as a 2
9	O03	How many crewmen are typically working on a fishing vessel?
10	O04	Do you want to know in advance when there are whirlpools around when you're fishing?
11	005	Would you like to receive advanced information on high and low tide?
12	O06	What types of alerts and alarms are received on communication equipment?
13	007	Would you like to receive information on the potential fishing zone (PFZ) while fishing?
14	O08	Do you want a technology that will inform surrounding fishing vessels in case of battery or engine failure?
15	009	What is the maximum distance and number of days spent in a fishing run?
16	O10	What is the average and maximum speed of the fishing vessel, and is there a difference in speed while fishing or searching?
17	011	What are the expected challenges and concerns with installing new communication equipment on the fishing vessel, and are you worried about your job?
18	012	What kind of real-time information do you need to share while fishing?
19	013	What kind of fishing vessel is used for fishing expeditions?
20	014	What are the strategies for maximising profit when selling fish at the shore?
21	O15, P01	How far are you getting cellular coverage at sea?
22	P02	Does your phone support 2G, 3G, or 4G cellular technology?
23	P03	Is it possible to have a video call with family while on a long fishing trip?
24	P04	Do you have the desire to communicate with family and others in coastal regions using specific instant messaging apps?
25	P05	What is the method for communicating emergencies to family and how can they reach you while at sea for fishing?
26	P06	What do you do during your leisure time while fishing at sea?

\*E - Emergency, O - Occupation, P - Personal.

(g) climate change impacts, (h) seasonal impacts on fishing, (i) perceived usefulness of existing solutions, (j) motivational aspects for digital solutions catering to entertainment, infotainment, business, etc. Through this inductive data analysis, a profound understanding of the community was achieved, identifying different problematic themes and establishing various categories to conceptualise a solution for empowering the fishermen's community. The insights from this stage were then applied to formulate the initial design of the OceanNet Digital Ecosystem (ODE).

#### S. Anand et al.

#### 2.3.3. Build and test

Before conceptualising and designing an actual solution for the pressing challenges identified in the community, an detailed literature review was conducted to identify available communication technologies at sea, along with business-related solutions, their feasibility, and challenges. The literature review on communication technologies specifically focused on those suitable for providing long-distance communication, as well as on blockchain technology for addressing businessrelated issues.

Furthermore, a review of the literature on various technology adoption methods and social change factors, which would play a pivotal role in the design process, was also conducted. Based on the identified requirements, a feasible technological solution was designed *(the ODE)*, and the communication solution was developed, as discussed in Section 4.2.1. The designed solution addressing communication issues, including components such as nodes and antennas, was implemented in a laboratory environment. In the subsequent phase, the research team tested the implemented solution to ensure it performed as expected.

#### 2.3.4. Deployment, training, and validation

The solution underwent an initial pilot deployment on three fishing vessels for extensive testing in the ocean environment with the fishermen's help. Researchers accompanied fishermen on these trips, providing guidance on system usage and assisting with rotating the vessel's antenna to establish connectivity. Careful consideration was given to the deployment of various modules on the vessels to minimize disruption to the fishermen's work. Following numerous rounds of testing, the solution's equipment was installed on an additional ten fishing vessels over a two-year period to assess user adoption. During this phase, fishermen with the installed equipment received comprehensive training sessions to enhance their understanding of the system's benefits. These sessions ensured user familiarity and efficient use of the equipment, contributing to the social change model detailed in Section 4.4. The specific training methods employed are as follows.

- 1. **In-vessel Training:** Researchers visited fishermen on their fishing vessels before they embarked on fishing trips. They explained and provided training to the vessel captain and other crew members on how to use the ODE communication system and mobile application, outlining the necessary steps to ensure the system functions effectively.
- 2. User Manual: Recognising the need for users to remember the steps for system operation, a user manual in the local language (Malayalam) was prepared and distributed to each fishing vessel for user convenience.
- 3. **Outreach Activities:** Several outreach activities were conducted to update and raise awareness about the developed solution and its advantages. This included visits to fisheries schools, to educate the new generation of the community, and outreach to fishermen's families as well as the self-help group members to provide a deeper understanding of the solution.

In the subsequent validation phase, three focused group [28] discussions, each with a minimum of ten participants, were conducted to gather feedback on the solution. This phase was crucial for refining the solution in the next stage.

#### 2.3.5. Update and improvement

Besides the sessions conducted during the validation phase, an additional six rounds of focused group discussions with users of the solution assisted the research team in continuously enhancing the solution. These discussions took place in various settings based on the situations: some meetings were held on the fishing vessels with the captains and crewmen, while more formal meetings occurred between the entire research team and stakeholders such as fishing vessel captains, business managers, boat owners, etc. Each discussion session involved between four to ten participants cooperating with the research team. The participatory approach also helped the team understand the additional requirements of the fishermen, focusing not only on technological aspects but also on user-friendliness and ease of use. The update and improvement processes were particularly crucial for the community to adopt the proposed solution.

During the pilot deployment phase, nine focused group discussions were conducted with fishing vessel captains to gather feedback on the challenges faced while utilising the proposed solution and to receive suggestions for making it more convenient for them. The close interaction with the stakeholders helped to improve the design on antenna placement, antenna rotation, finding a suitable location for the controller box inside the boats' wheelhouse, and efficiently wiring the required cables. In the re-design process of the solution, the team benefited from more frequent focused group discussions with selected fishermen and fishing vessel owners who participated in the initial deployment and testing phases. These focused group discussions also aided in assessing the community's digital literacy, digital capacity, and alternative skill sets. Fig. 4 illustrates the methods employed by the research team during the various phases of the study. It also depicts the achieved objectives of each method. The figure delineates the process sequentially, starting from gathering information about the community to delivering a solution based on the identified requirements. It provides a different perspective on the ODE development, showcasing where in the process digital capacity-building was delivered.

## 2.4. Sustainable social change

This study complies with the three pillars of sustainable development, environmental, economic and social pillar from different perspectives [36]. To comply with the environmental pillar, the reliable and transparent data records on the blockchain database (ledger) enable the policymakers and environmental activists to have an accurate picture of the exploitation of the marine ecosystem at any time so they can make informed decisions when needed. The ODE significantly bolsters the social aspects of sustainable development through several key strategies. First, it empowers fishermen by reducing their vulnerability to exploitation by middlemen, fostering a more equitable fishing environment. Second, ODE prioritises active engagement and participation from the community, recognising their role in its success. Knowledge sharing is also central to ODE's approach, as it actively works to improve knowledge and awareness among community members. Furthermore, social inclusion is a core principle throughout the entire co-design process of ODE. Ultimately, by empowering fishermen, ODE contributes to increased livelihood resilience for the entire community. Additionally, the system has the potential to reduce health risks encountered by fishermen during voyages.

Following an in-depth examination of the community's challenges and an analysis of their unique needs and characteristics, a comprehensive suite of technological solutions has been developed. This digital ecosystem is tailored to facilitate their digital empowerment. To ensure the sustainability of the proposed digital ecosystem, it was crucial to cultivate motivation for technology adoption within the community and enhance the digital proficiency of stakeholders, such as fishermen. This would enable them to effectively utilise the system to enrich their livelihoods and ensure safety during their voyages. The sustainable social change model applied in this study comprises two main sets of strategies. The first is aimed at motivating the community to accept the necessary social change, while the second focuses on enhancing their capacity and confidence to adopt and utilise the proposed solution for addressing their safety and livelihood issues. The methods employed to boost motivation included.

• Social Influence: Social influence, broadly defined as the impact of others on our behaviour and choices, is a key factor in technology adoption among the fishermen. In this case, it includes their



Fig. 4. Methods used for requirement collection and digital capacity building based on the different phases of the study.

perception of their peers' technology use [37,38]. This project leveraged social influence to enhance the motivation of fishermen by involving family members, peers, and community leaders. Community leaders were particularly effective in persuading fishermen to try the proposed system during their voyages. To further bolster trust and encourage adoption, the highly respected chancellor of the university, who himself hails from the fishing community, played a key role in communicating with local leaders and fishermen about the system's purpose and capabilities. Additionally, several other researchers and staff from the university, with roots in fishing communities, were involved in encouraging stakeholders to utilise the proposed technological solution.

• Incentivisation: Researchers provided incentives such as real-time communication while at sea, free Internet access, access to market information for informed decisions, and availability of infotainment to encourage stakeholder system usage. Additionally, the research team conducted video calls from the ocean to the fishermen's family members on shore to boost their motivation. This was especially impactful since the fishermen typically have no means of communication with their families for an average of a week while they are at sea.

Studies show that a community's belief in their ability to use a technological solution – possessing the necessary skills, knowledge, and tools – is crucial for adoption and utilisation [38]. Furthermore, sustained use fosters habit formation and reduces user anxiety, ultimately increasing the likelihood of long-term adoption [38]. The research team focused on enhancing the users' mental capabilities and technical proficiency to achieve this sustained adoption of the proposed digital ecosystem.

Methods utilised to enhance these capabilities and proficiency in the required skill set of the users include.

- Educate: Researchers conducted several awareness sessions on the benefits of technology use and the advantages it brings to fishermen, their families, and vessel owners. Researchers have done several vessel visits and one-on-one discussions with fishermen on the benefits and incentives they receive from learning how to utilise the system. The interaction was also used to understand their underlying beliefs, the skill sets to be built to operate the system, and the proper utilisation of the technology.
- Train: To ensure stakeholder proficiency, training sessions were conducted in two phases: onshore before fishing trips and offshore during voyages. These sessions focused on equipping participants with the skills to operate the communication equipment on their vessels and utilise the OceanNet mobile application on their smartphones. The researchers meticulously covered a range of topics, including operating the communication device, remotely controlling antenna rotation, interpreting signal strength readings, identifying optimal antenna positioning times, recognising base station locations, and maintaining the system onboard. These comprehensive training sessions were repeated multiple times before the fishermen embarked on independent fishing trips, ensuring they could operate the technology without researcher accompaniment.
- Build Confidence: Training fostered a supportive environment where researchers were impressed by the fisher-men's eagerness to learn the new technology. They actively encouraged independent use of the communication equipment and software, assuring participants of ongoing support whenever needed. To further empower the fishermen and address any technology anxiety, researchers embarked on several fishing trips alongside them. During these voyages, the fishermen were guided through using the system for communication, including services like Skype, social media, and instant messaging. Researchers provided clear demonstrations and explanations, ensuring the fishermen felt comfortable and confident utilising the technology.

The result of this social change model implementation for the

adoption and usage of the technology, their feedback and the detailed observed changes in users over several months have been explained in section 4.4.

#### 3. Literature review

Bridging the digital divide remains a significant challenge for empowering coastal fishing communities. Several factors contribute to this challenge, including the lack of long-range communication systems, solutions for inclusive stakeholder engagement, and successful interventions to enhance technology adoption. This section delves into these key challenges and explores existing literature to identify multidimensional factors and critical research gaps.

The digital divide refers to the disparities in Information and Communication Technology (ICT) access, usage, and outcomes, posing a major societal challenge [39]. This issue significantly hinders the sustainable development of communities. The digital divide can be addressed at three key levels [40].

- Level 1: Digital Access and Internet Attitudes: This level focuses on disparities in access to the internet and the general attitudes towards its use within a community.
- Level 2: Digital Capacity, Skills and Usage: This level deals with the skills gap related to technology use and the actual use of technology within the community.
- Level 3: Beneficial Digital Outcomes: This level addresses the disparities in the positive outcomes derived from technology use within a community.

#### 3.1. Digital access

The lack of digital connectivity has a significant impact on the economies of rural communities, particularly regarding communication at sea and its effect on the competitiveness of small-scale businesses [41]. Fishermen rely on various communication technologies depending on the communication scenario: (a) connecting with families or vessel owners on shore; (b) communicating with nearby "buddy vessels"; and (c) long-distance emergency communication. Unfortunately, no single affordable technology effectively addresses all these needs. For communication between buddy vessels at sea, fishermen currently utilise VHF (Very High Frequency) and UHF (Ultra High Frequency) radio communication systems [8,42]. These technologies are suitable for short-range communication, typically within a few kilometres of the shore, but are susceptible to interference from various sources like other radios, electrical equipment, and even the metal construction of the fishing vessels themselves. Additionally, the Earth's curvature and obstacles like the sea surface can limit their range. Furthermore, bad weather conditions such as heavy rain or lightning can disrupt their performance, rendering communication unreliable or even impossible [43,44].

Existing solutions for communication with families and boat owners on shore, such as 4G/5G cellular networks, are limited by range, typically reaching a maximum of 15 km from the coast [45]. Cellular operators are hesitant to invest in expanding coverage due to the low population density in these coastal villages, resulting in a poor return on investment (ROI) [45,46]. While technologies like maritime satellites, the Automated Identification System (AIS), and Starlink offer high-throughput communication capabilities, their high costs render them unaffordable for many fishermen. Additionally, Starlink satellite services face limitations in remote areas due to the high cost of the solution and, as of December 2023, regulatory hurdles [47]. These regulatory challenges, coupled with the need to install user terminals (UTs) on fishing vessels, further restrict the feasibility of Starlink for a significant portion of the fishing community. potential solutions, acting as relays or aerial base stations to connect to satellites [48]. However, the high costs associated with purchasing, operating, and acquiring the skills necessary to operate these drones for communication purposes pose a significant barrier. Furthermore, bad weather conditions can disrupt their functionality, and commercial use of drones often requires special licenses or permits. Since typical fishing voyages exceed 100 km, existing affordable technologies struggle to provide connectivity beyond cellular coverage areas. This results in a communication blackout for several days during each fishing trip, which contributes to the heightened risks associated with being at sea. Consequently, there is a pressing need for the development of low-cost, low-power, long-range communication systems that enable real-time communication for fisherfolk communities [8].

#### 3.2. Digital capacity and usage

Research indicates that digital technology use can improve people's quality of life [49]. However, the digital divide creates inequality in Information and Communication Technology (ICT) use within rural communities. Digital capacity, the second level of the digital divide, refers to the skills required for tasks such as information searching, using collaborative software like WhatsApp and Skype, basic internet navigation, and overall ICT experience and use [50]. These digital skills are crucial for maximising the benefits of available services and technologies. Unequal skills and knowledge in handling internet devices are termed 'digital inequality'. This inequality encompasses a user's autonomy (referring to the user's ability to make independent choices regarding the use of digital technologies), usage patterns (meaning the frequency of digital device usage), and overall skill level. Additionally, factors such as low self-efficacy, computer anxiety, and other psychological influences are considered to contribute to digital inequality [39].

Motivational access, another key factor, refers to the desire to have a device and connect to ICT [51]. Similar to the previously mentioned factors, motivational access inequality can hinder users from adopting specific technologies. Motivational access is considered to be influenced by social, cultural, mental, or psychological factors, and can be rooted in early social experiences that may make the internet seem unappealing to low-income and low-educated individuals [52]. The motivational access divide includes a lack of motivation to access ICT, learn new technologies, or have any interest in IT-related topics. However, even with limited physical access to ICT, high motivational access can encourage individuals to develop stronger skills in using digital technologies [53].

Varma et al.(2021) pointed out that the critical challenge in designing a solution is to engage the community freely, understand the challenges faced by the community, such as water scarcity in the summer, and then decide on appropriate technological solutions [54]. They have used a participatory co-design approach for enhancing digital capacity, and this method is used to identify and solve their water crisis collectively, involving ice-breaking sessions, mind maps, group sketching, and brainstorming. The study conducted by Ajith et al. (2022) pointed out the need for a phased approach for implementing the Jivamritam - drinking water solution in rural areas, through capacity development and technology transfer [55]. The study addresses challenges such as imparting knowledge, resource allocation, technology selection and provisioning, community empowerment, water governance, and local government support. They addressed these issues by establishing a Community Empowerment Model through engagement, co-design, and ownership at individual, collective, and practical levels, aiming for a sustainable outcome of pure drinking water for the community. Their interventions lead to raising awareness, designing a user-friendly solution with minimal interventions, instilling a sense of ownership, and also making them aware of the solution's constraints and cost implications.

UAV (Unmanned Aerial Vehicle) drones have also been explored as

## 3.3. Digital outcome and technology adoption

According to the International Telecommunication Union (ITU), the assumed outcomes of digital skills for the global population are the ability to use ICTs to achieve favourable outcomes in the day-to-day life of themselves and others [56]. However, the outcomes in one domain may not necessarily correlate with outcomes in another, highlighting the need for more research to draw conclusive results from existing studies and their impact on intended outcomes [56]. Additionally, inequality in users' capacities to exploit ICT can significantly affect individual outcomes and benefits [39]. For instance, a study focusing on students' performance in rural and urban communities in the context of e-learning revealed that the difference in the digital divide is mainly due to a lack of parental support and intrinsic motivation [40]. This lack of support and motivation decreases the intended benefit of ICT, affecting the digital outcome. In another study, a community engagement approach based on the theory of change was successfully adopted, using peer educators as mediators of knowledge transfer to create a supportive environment at both household and community levels for a better digital outcome [57].

Adopting digital solutions and moving toward digitisation play a crucial role in achieving sustainable development goals across multiple sectors [58]. The lack of knowledge or skills to use technology leads to poor adoption of technology [41]. According to the research by Al-Emran et al. (2023), adopting technology is not an immediate process, as it requires non-technological behaviour changes [49]. The study presented by Hwang et al. (2023) explores the roles of critical factors based on social cognitive theory such as social, personal cognitive, and the commitment to change approach in e-vehicle adoption by the community [59]. This study highlights that social transformation requires a change in individual perceived motivation. However, individuals often exhibit inertia in adopting new services or products and tend to stick to familiar ones. The study also reports that resistance to change, lack of understanding, and trust issues act as significant barriers to the adoption of AI in the supply chain area [60].

An extended version of the Technology Adoption Model (TAM) for mobile health services has been proposed [61], incorporating six new variables. These variables encompass technology anxiety, social influence, trust, perceived risk, perceived physical condition, and society's resistance to change. Technology anxiety pertains to the extent individuals experience problems while adopting a new technology [61]. Social influence plays a significant role in shaping the decision to adopt the technology and also impacts the perception of usefulness (PU). Research has shown that social influence fosters positive attitudes and behavioural intentions. For instance, in a study by Yu et al. (2017), it was found that observing individuals within one's social circle using an interactive communication tool strongly motivated others to adopt the same ICT tool [62]. Social innovation studies recognise innovation as a driving force for societal improvement and positive change [63]. Additionally, social influence plays a crucial role in technology adoption, enabling individuals and society to embrace new technologies [64]. Consequently, Internet and communications technologies are widely embraced in individuals' daily lives due to social influence.

Creating new technologies for low-literacy groups requires in-depth research into their lives, relationships, and concerns, and a continuous partnership with primary users. For mobile technology targeting older people, consideration of their physical and cognitive profile is essential [65]. Involving users throughout the design process allows for progressive testing of the solution and facilitates fine-tuning of the system, thus strengthening the design [66]. One of the major barriers to the success of any new information technology solution is the lack of user acceptance [30,67,68]. User acceptance refers to the readiness of a user group to use Information Technology (IT) for the task it is meant to support [69]. Therefore, user acceptance is considered a crucial factor in deciding the success or failure of any information system project [30, 70].

been discussed in several studies [68,71–75]. Among the key factors contributing to user acceptance of technology is the user's confidence and the ease of use of the product [76]. Previous research indicates that users are sometimes hesitant to adopt information systems that promise impressive performance gains [77]. Researchers have explored a wide range of issues related to this topic, covering individual user characteristics, their cognitive styles, internal beliefs, and impact on user behaviour [78]. User acceptance has also been viewed as a function of user participation in system development [79]. Perceived Ease of Use (PEU) and Perceived Usefulness (PU) play significant roles in developing attitudes towards technology adoption [67,80–82]. PEU encompasses various factors, including the user's self-efficacy and the complexity of the interface [83]. The impact of technology can be measured based on its acceptance [83].

Studies have demonstrated that digital habits and digital skills help to move with the current digital era, and they have the potential to transform the community [84,85]. Because digital transition impacts humans on their cognitive level, individual level, and social level [84], Internet skills and experience even protect people from digital exclusion. The current digital technologies are changing individuals' social, behavioural, and cognitive development due to their highly digital habits [85]. Additionally, technologically enabled social structures are emerging, facilitating fundamental changes in business practices.

Another study explored different factors that affect the user's intention to adopt new consumption behaviours required for adopting smart grid technologies through the use of the Theory of planned behaviour (TBA) framework, which is an extended model of the Theory of reasoned action (TRA) [86]; both models affirm that individuals' behavioural intention generally leads to their particular behaviour. Similarly, any change imposed on users can induce resistance, and it is a normal response to any innovation. The study also suggests that behavioural beliefs and attitudes can increase positive attitudes and facilitate adoption, as attitude is a factor that leads to adoption intention. The study suggests three interventions: education, communication, and information initiatives. The findings highlight that communication and information initiatives must focus on content related to specific beliefs [86].

A study reported factors influencing the acceptance of telemedicine services by the rural population of Pakistan. They have used TAM as the framework and derived factors linked to the usage intention [87]. The research team adopted a face-to-face interview method and used the Partial Least Squares (PLS) method to analyse the collected data. Research suggests that usage intention is a function of several factors, including PU, PEU, social influence, facilitating conditions, and factors such as trust, technological anxiety, perceived risk, and resistance to technology [87]. Another research focused on adoption intention, highlighting the importance of three types of access, mental access, material access, skill and usage access as the significant contributing factors towards the adoption of ICT among rural women [88]. They point out that adopting ICT can lead to innovation, and adoption intention will boost their entrepreneurial orientation. They used a Structural Equation Model (SEM) to validate and test the model.

Furthermore, in addition to considering the perspective of technology adoption, it is crucial to take into account the dimensions and factors necessary for facilitating social change towards technology adoption. According to the literature, there are several dimensions to social change such as the pace of change [89], the scope of change [90], the direction of change either progressive [91] or regressive [92] directions, the level of disruption, revolutionary [93] and evolutionary (gradual) [94], the intentionality of change, planned and unintended changes [95,96], and distribution of benefits of change [97] to name a few. There are also several key factors to social change realisation such as education level, social influence, and capacity building [98–100], time and duration of engagement with the community, participative action, and local social networks [101,102], human tendency to change [99].

# 3.4. Identified research gaps

The key research gaps derived from the detailed analysis of the existing literature are.

- Limited long-distance communication solutions hinder real-time offshore connectivity for fishing vessels, restricting their digital access and hindering the full potential of digital technologies for these communities.
- Existing research on digital inequality often overlooks the specific challenges faced by fisher communities. A more in-depth assessment is needed to understand the factors that hinder digital capacity building within these communities.
- A critical gap exists in understanding how enabling access to tailored digital services can enhance technology adoption for improved safety and livelihood outcomes.
- The literature extensively explores how technology shapes social structures. However, a gap exists in investigat-ing the role of community stakeholders as technology adopters in this process.

This section presents the findings of the study, including the analysis of data collected from the coastal fishing community. It details the architecture of the designed digital ecosystem, which aims to empower fishers through digital access and the ability to utilise digital services for enhanced safety, communication, and livelihood opportunities. Finally, the section explores the implementation of the proposed solution and the sustainable social change model employed in the study.

#### 4. Results

#### 4.1. Data analysis outcome

The community in this area is more susceptible to ocean-based disasters such as coastal floods and tsunamis [103]. The local people in this coastal region exhibit distinct behaviours, norms, practices, and cultural characteristics, reflecting their unique relationship with the ocean environment. They frequently face the unpredictable behaviour of nature, leading to significant challenges and hardships. During the initial assessment of the community's engagement process, the research team discovered the profound digital divide specific to the fishing community in Kerala, India. Most of the fishermen and their families are not digitally empowered to use the Internet, smart devices, and applications for their daily communication, information retrieval, and entertainment. Several factors such as mixed level of education, lack of smartphone usage, lack of interest in ICT, technology anxiety, and lack of digital skills contribute to this divide, with the primary issue hindering the safety of fishermen at the ocean being the lack of communication and internet facilities beyond a certain distance from the shore. Existing communication opportunities are limited to only a few kilometres or are economically infeasible for the fisher community. Some other solutions may require special licenses, and many are not viable in the harsh ocean environment. The detailed findings from the conducted interviews with the community members are as follows.

• Fishing vessel operations: The face-to-face interview findings revealed that fishermen typically embark on fishing expeditions, sometimes lasting for up to fifteen days and, in rare cases, up to one month, depending on their catch and available fuel. These journeys involve a crew comprising the captain or 'srank' (driver), and workers assigned and tasked with capturing fish using the net and sorting various catches and bycatches. It was observed that a few vessels have a single individual who fulfils the roles of both captain and driver. The decision-making process during fishing operations rests primarily with the captain, including trip duration and contacting the manager or seller at fish landing centres. Additionally, they are responsible for navigation, locating fish types, and communicating

with other fishing vessels using the VHF (Very High Frequency) radio within short distances. Usually, the crew size varies between 10 and 12. The size of the fishing vessel also varies between 80 feet and 110 feet. There are two types of fishing vessels, one made of wood and fibre and the other one made of metal. They fish below the Arabian Sea area and the fish gears they use are primarily of two types such as *bottom trawl* and *midwater trawl*. However, based on the mouth opening of the trawl they are grouped into beam trawl, otter trawl, and bull trawl.

- Communication challenges: The face-to-face interview findings revealed that fishermen typically embark on fishing expeditions lasting up to fifteen days, and in rare cases, up to one month, depending on their catch and available fuel. These journeys involve a crew comprising the captain, also referred to as the 'srank' (driver), and workers tasked with capturing fish using nets, sorting catches, and bycatches. It was observed that some vessels have a single individual fulfilling both captain and driver roles. The decision-making process during fishing operations rests primarily with the captain, including trip duration, contacting managers or sellers at fish landing centres, navigation, locating fish types, and communicating with other fishing vessels using Very High Frequency (VHF) radio for short-distance communication. Crew size typically varies between 10 and 12 members. Fishing vessels range in size from 80 to 110 feet and are constructed from either wood and fibre or metal. The primary fishing area is the Arabian Sea, and the main fishing gear types are bottom trawls and midwater trawls. These trawls can be further categorised into beam trawls, otter trawls, and bull trawls based on the mouth opening design.
- Smartphone usage and technology gaps: The study revealed that the majority of fishermen operating fishing vessels are middle-aged or older men who are unfamiliar with the benefits of using smartphones. Currently, smartphones are primarily used for making voice calls and sending text messages. Among the younger generation of fishermen, some carry smartphones, but these are often encased in plastic bags once they reach the limit of cellular coverage, as the phones become unusable beyond this range. The interviews further indicated that most fishing vessels lack Automatic Identification System (AIS) support or satellite communication due to the high costs associated with acquiring and operating such technology. Fishermen expressed a strong desire for alternative communication methods to ensure more reliable and efficient communication at sea. Additionally, a few fishermen expressed interest in using communication methods to connect with their families while at the ocean, with children in their families particularly keen on maintaining communication during their fathers' absences.
- Digital divide and lack of access to digital services: The study highlighted the fishermen's struggle to access information from the Indian National Centre for Ocean Information Services (INCOIS), particularly regarding Potential Fishing Zones (PFZs) and other relevant alerts. This lack of connectivity often results in receiving such information too late to be practically useful. Furthermore, the research revealed that many fishermen expressed a need for awareness about the digital services provided by INCOIS, including weather forecasts, high sea wave alerts, temperature updates, and fishing zone information. The absence of connectivity at sea creates a distinct lifestyle for these fishermen compared to nearby villages. The majority of fishermen lack experience using smartphone applications such as internet banking, mobile banking, UPI transactions, instant messaging, and social media for connecting with family and friends. Similarly, they have no prior experience using safety-related services like INCOIS alerts and other weather information. This lack of digital access creates a sense of isolation from the outside world and contributes to a different way of life compared to those residing across the lagoon that separates this community.
- Secrecy and competition: During a closed group discussion, a fisherman emphasised the need for confidentiality regarding certain

fishing information, such as the location of identified fish schools, to avoid competition with other vessels. VHF channels currently used for communication present a challenge in maintaining secrecy, as others can easily overhear such information. The fishermen also highlighted the difficulty in conveying catch information to fishing vessel managers in a timely manner. This information is crucial for facilitating the selling process to third-party buyers in fish landing centres, where auctions determine the final price. The lack of timely communication between managers on shore and fishermen at sea limits their ability to capitalise on these auction opportunities. Furthermore, the absence of real-time information on fish market demand from the shore makes fishermen entirely reliant on intermediaries for the fish-selling process. This reliance can negatively impact their profit margins. Access to communication and internet connectivity would empower them to directly communicate critical information with various stakeholders throughout the process, potentially increasing their profits. The fishermen in this area typically catch over 30 different fish species. Some of the most common and commercially valuable catches include various types of Sardinella, Mackerel, Seer fish, Black Pomfret, Tuna varieties, Pink Perch, and Indian Anchovy.

- *Community structure and influence*: The various stakeholders involved in fishing, from processing and selling to distribution, exhibit a strong interconnection, working collaboratively to sustain their livelihoods. This collaborative spirit extends beyond the immediate community, as evidenced by the shared use of fish landing centres with nearby villages. A strong sense of community is evident, with daily interaction among members and engagement with nearby villagers to make occupational decisions.
- *Safety concerns*: Safety emerged as a paramount concern during fishing expeditions. Fishermen reported facing significant risks from large cargo ships traversing their fishing grounds and resting areas, potentially leading to accidents and chaotic situations. The current AIS and VHF systems appear inadequate in effectively preventing collisions or providing timely warnings. Compounding this issue, many fishing vessels lack AIS entirely, leaving them vulnerable to the lack of emergency notifications during situations like cyclones, due to the absence of connectivity. Fishermen further highlighted recurring concerns such as vessel engine troubles, fuel depletion, capsizing incidents, and unintentional straying into shipping channels, all of which pose significant risks.

Table 2 outlines the specific digital access needs identified through interviews and other methodologies. Addressing these needs and providing suitable technological solutions has the potential to

#### Table 2

Digital	access	require	ments	of the	community	
Digital	access	require	memo	or the	community.	

0		
Emergencies	Occupational	Personal
<ul> <li>Real-time localisation and tracking for rescue</li> </ul>	- Sea state information	<ul> <li>Daily phone calls or messaging</li> </ul>
<ul> <li>Remote health monitoring for emergency in case of illness and injury</li> </ul>	- Potential fishing zone	<ul> <li>Internet and infotainment solutions with adequate bandwidth to the shore</li> </ul>
- Critical weather, sea information and disaster alerts	- Disseminate fishing market price information for auction	- Emergency information contact
- Communication with port authorities	<ul> <li>Disseminate fuel and fish storage information</li> </ul>	<ul> <li>Personal communication to the crew members of other fishing vessels</li> </ul>
<ul> <li>Country border crossing alert</li> </ul>	<ul> <li>Navigational and mapping information</li> </ul>	- Communication with family members
<ul> <li>Peer-to-peer vessel communication</li> </ul>	<ul> <li>Peer-to-peer vessel communication</li> </ul>	<ul> <li>Communication with fishing vessel owner</li> </ul>
- Ship to fishing vessel collision	- Logging of fish catch	- Communication with fishing vessel manager

significantly improve the fishermen's efficiency, safety, and overall fishing experience. The findings from these methodologies form the foundation for designing and developing the proposed digital solution. Recognising the existing digital divide, a co-design approach was employed to create the solution. The objective is to empower the fishing community by enhancing their livelihood opportunities and resilience while bridging the digital gap. This research introduces a comprehensive integrated digital ecosystem called the OceanNet Digital Ecosystem (ODE). This digital ecosystem is meticulously tailored to meet the digital access, digital capability, and digital outcome requirements of the fishing community. Ultimately aiming to foster sustainable and resilient fishing communities. The following section elaborates on the various aspects of this unique digital ecosystem.

#### 4.2. Architecture of the proposed digital ecosystem

To empower the fishing community by enhancing their livelihood opportunities, and resilience, this research proposes a unique integrated digital ecosystem called the OceanNet Digital Ecosystem (ODE). The significant regional variations in digital service demand and supply highlight the importance of effective digital ecosystems [104]. The ODE integrates three key solutions to enhance safety at sea and strengthen community resilience.

#### 4.2.1. Internet over ocean: a hybrid communication network

During fishing voyages, fishing vessels travel vast distances remain at sea for extended periods and lose connectivity from the shore. To overcome this challenge, the solution leverages the fishing vessels themselves as relay nodes in the ocean, effectively extending the communication reach. To achieve this, a heterogeneous multi-tier ad hoc network architecture has been integrated to expand the limitations of existing communication networks and establish the foundation of the OceanNet solution. The OceanNet communication system utilises longrange Wi-Fi to establish both Vehicle-to-Infrastructure (V2I) and Vehicle-to-Vehicle (V2V) networking, enabling network access deep within the ocean. Fig. 5-(a) illustrates the network architecture of the fishing vessel communication system. Based on a detailed technical evaluation [45], Long Range Wi-Fi (LR Wi-Fi) was chosen to provide backhaul internet connectivity, making the V2I network possible. Wi-Fi technology is also used for communication between fishing vessels via V2V networking, allowing fishermen to access the internet through their smartphones.

The implemented network architecture consists of three distinct tiers, each offering varying communication capabilities and incurring different installation costs. This tiered structure allows boat owners to select the equipment that best suits their financial situation and vessel size. The Supernode (SuN) tier represents the most resource-intensive node configuration, featuring an AR and two ABEs installed on the vessel. The Adaptive Node (AdN) tier equips vessels with both an AR and Adaptive Backhaul Equipment (ABE). While the ABE is more expensive than the AR, it offers a significantly extended communication range. The last one is the Access Node (AcN) tire, which utilises a relatively inexpensive Access Router (AR) installed on smaller vessels. The AR functions as a Wi-Fi router with a communication range of 100-200 m. All these nodes leverage extended communication capabilities by integrating V2V and V2I networking concepts. The V2V network provides internet connectivity among fishing vessels in the ocean. V2I communication is established between SuNs and AdNs, while V2V communication occurs between AcNs and both AdNs and SuNs. The V2I network provides offshore connectivity by strategically placed base stations with elevated heights. These elevated base stations offer improved line-ofsight and extended communication range. Automated antenna reorientation systems and directional beamforming techniques are employed to continuously track the location of the base station for optimal connectivity. The dynamic and evolving nature of fishing vessel locations creates a hybrid, self-organising network topology, facilitating



Fig. 5. (a)OceanNet network architecture for fishing vessel communication [105] (b)Web Application for localisation and tracking.

extended communication coverage across the ocean.

#### 4.2.2. Mobile application for real-time fishing vessels tracking

Community engagement revealed the need for real-time vessel localisation and tracking when fishermen are at sea. Additionally, the fishing community expressed a strong desire for access to information on potential fishing zones (latitude and longitude information of fish availability) provided by INCOIS while at sea and showed interest in voice-over-IP calls, information gathering, and entertainment. All these user needs and functionalities have been integrated into the upgraded OceanNet Digital Ecosystem (ODE). This was specifically designed to foster motivational access, aiming to generate increased interest among community members in utilising the ODE.

Recognising the importance of perceived usefulness, the system prioritises real-time vessel tracking. A device installed on each vessel continuously transmits its GPS location, enabling the monitoring of the vessel's live position and associated connectivity parameters as shown in Fig. 5-(b), allowing visualization of a vessel's location and distance from shore. The web application user interface (UI) leverages the Open-StreetMap API to visualise these real-time location updates. Currently, both fishermen's families and vessel owners can utilise the software to track the live location of their vessels. Incorporating user feedback led to the development of characteristic features like icon-based design and multilingualism in later stages. This mobile application serves as a significant driver for digital skill development and motivational access.

# 4.2.3. Blockchain-enabled e-commerce platform for livelihood enhancement

Fishing communities face constant challenges from climate change (cyclones, floods, tides) and disasters (tsunamis). The 2004 tsunami devastated the Alappad community in India, causing social and economic hardship, highlighting the need for livelihood support [106]. Given the susceptibility of the fishing livelihood to climate change and market swings, building resilience is crucial to absorb these shocks and prevent further hardship [107]. The current fishing livelihood faces several challenges that limit their income and decision-making power.

The lack of communication and traditional first-come-first-serve auction system demands a physical presence at the port, limits their role in the process and makes them vulnerable to non-inclusive practices. The lack of digital skills and digital literacy restricts access to broader, potentially more profitable markets. Also, the current fishing industry lacks transparency and control for fishermen. Middlemen dominate the process, influencing prices, limiting market access (due to trust issues and reliance on intermediaries for information), and restricting fishermen's roles in decision-making and auctions. This dependence weakens their livelihood security and limits their ability to adapt to market fluctuations and access a wider customer base. A solution is needed that empowers fishermen by fostering inclusivity in decision-making, providing real-time market information, eliminating the need for intermediaries, and allowing fishermen to access a wider market while at sea.

To address these requirements, a blockchain-enabled solution has been devised to enhance engagement, inclusivity, and profit for fishermen. A blockchain-based digital bidding process (digital auctioning function) has been proposed to transform the centralised, middlemencontrolled auctioning process by providing crucial market fluctuation information on a tamper-proof blockchain ledger, ensuring high reliability and transparency [108-110]. All transactions will be managed by smart contracts through a decentralised mobile application (DApp). Unique digital identities track the sellers' and buyers' activities over the platform. Instead of cryptocurrencies, existing digital payment systems like UPI and bank transfers will be utilised for a better user experience, connecting with a wider range of customers and enhancing profit margins and livelihood opportunities. This approach mitigates over-dependency on intermediaries. Fig. 6 depicts the benefits of the proposed solution and the place of improvement in an auctioning process (green boxes).

The study revealed that several fishermen lack sufficient digital literacy or financial means to own smartphones. To overcome these challenges, dedicated smart warehouses need to be established at each harbour to facilitate trades via a digital platform [11]. Operators of these smart warehouses, known as Oracles, will input data about available fish products into the blockchain, shaping the supply chain management system for the fish market. The oracles can be selected from among the women within the community, promoting gender equality. The proposed solution provides fishermen with Internet access through a centralised approach and offers a decentralised digital market and supply chain management system, creating a hybrid design serving users via a smartphone.

In summary, the proposed blockchain-enabled e-commerce platform offers numerous benefits to fishing communi-ties. It revolutionises the traditional auction process, empowers fishermen to conduct real-time auctions at sea, reduces dependence on intermediaries, ensures transparency, and eliminates exploitation risks. The digital ecosystem provides comprehensive information on fish product availability, prices, and climatic conditions, enabling stakeholders to make dynamic and informed decisions. The integration of smart warehouses as Oracles enhances information inclusivity and shapes the supply chain management system, benefiting the fish market.

#### 4.3. ODE implementation

The core of the project's design and implementation revolved around enhancing the community's digital capabilities and overall digital well-



Fig. 6. The auction process, after the intervention via ODE.

being through an IoT-based digital ecosystem. Emphasis on solution acceptance, frequent usage, and ongoing community interaction was prioritized from the outset. During the community analysis phase, the research team conducted a comprehensive assessment to understand the stakeholder landscape. This included evaluating their current level of digital access and capacity, existing business practices, identification of influential stakeholders, and understanding the personas and challenges faced by the fishermen. This holistic approach ensured the solution would be aligned with their actual needs. Furthermore, the initial investigations also focused on identifying the various factors that contribute to the community's vulnerability to ocean-related disasters. Frequent engagement with community stakeholders proved instrumental in highlighting the need for a sustainable social change model. To ensure successful adoption of the OceanNet digital ecosystem, the project implemented capacity-building programs tailored to the community. These programs included in-vessel training sessions, the development of user manuals, and the strategic involvement of women's groups within the community.

## 4.3.1. Deployment

The solution deployment unfolded in stages, acknowledging the need to persuade the community of the system's benefits. Initially, three fishing vessels were equipped, targeting owners who recognised the value and importance of a vessel tracking system. The second phase expanded to four additional vessels, with referrals coming from satisfied owners and captains involved in the initial deployment. Positive feedback from these early adopters, particularly regarding instant messaging apps and voice/video call services, fueled discussions within the fishing community. This positive word-of-mouth ultimately led to the solution's adoption by the newly included vessels. Over the following year, the project expanded once more, bringing the total number of equipped vessels to fourteen.

The majority of vessels in this project are mid-sized, under 30 m long. Representing a significant financial investment for owners, the potential loss of a vessel can be financially crippling. Recognising this reality, some owners adopted the proposed vessel tracking system to enhance safety and security at sea. They perceived the system's ability to track vessels during routine voyages and provide crucial location information in case of emergencies.

Fishing vessel owners play a pivotal role in driving change, as their investment is on the line and they hold the ultimate decision-making power regarding system implementation. However, even with owner buy-in, securing the fishermen's active participation remains crucial. Several factors contributed to initial resistance among some fishermen, including concerns about privacy, technology anxiety, and the perceived complexity of using smartphones and the system itself, particularly among older crew members.

#### 4.3.2. Field trials to validate the architecture

To test and evaluate the communication solution and its architecture, testbed-based experiments were conducted over the Arabian Sea. The equipment were integrated into a total of fourteen fishing vessels for pilot testing as shown in Fig. 7.

The experiments were conducted under various Douglas sea states (a system for categorizing wave heights and sea conditions) [111]. Table 3 and Fig. 8 depict the range at which communication was successful: over 50 km for V2I communication (vessel to shore station) and over 15 km for V2V communication (vessel to vessel).

The pilot test phase achieved a successful communication range of over 65 km between a vessel and the shore using the designed communication architecture. This performance held true even under extreme weather conditions. Currently, the system is implemented and functioning well in fourteen fishing vessels, with real-time data collection enabling the monitoring and analysis of fishing vessel activity [45].

Initially, the solution relied on a manually operated rotating antenna to establish communication with the shore. Since the base stations utilised directional antennas, optimal connectivity required the captains to physically rotate the vessel's antenna towards the base station. The research team trained the captains on this procedure, and the maximum achievable angular connection range was approximately 17 km. In response to user feedback, the system was upgraded to incorporate a remote-controlled rotating antenna system. This significantly improved user experience and ease of operation. To further simplify antenna positioning, a fully automated antenna rotation platform was designed and integrated into the solution. This upgraded system utilises GPS location data to automatically adjust the fishing vessel's antenna orientation towards the shore station, eliminating the need for manual or remote control. Upon gaining internet access, the fishermen explored various digital services. These included messaging applications like WhatsApp and voice/video calls through platforms like Skype. Additionally, they utilised web browsers like Google Chrome to verify their connection quality. Twelve fishermen participated in this testing phase,



Fig. 7. Testbed setup for the network architecture: (a) Base Station setup at the shore, (b) Fishing vessel as a node,(c) SuperNode acts as mobile BS and ABE, (d) Adaptive Node or ABE.

Table 3					
Summary of six field	trials conducted a	at the Arabian	Sea during	high t	ides.

5				0 0					
Field trial date	Frequency (GHz)	# of Trawlers	Douglas seastate code (0–9)	Range (km) (shore to fishing vessel)	Range (km) (fishing vessel to fishing vessel)	Noise floor (dBm)	Signal strength (dBm)	Channel bandwidth (MHz)	Minimum Tx/Rx rate (mbps)
25/12/ 14	5.8	1	3	17.7	-	-92	-73 to -88	5	1.725/1.625
01/06/ 15	2.4	2	3	45.6	16	-99	-58 to -86	5	3/3
06/01/ 16	2.4	3	3	43.7	22.6	-98	-62 to -82	5	4.9/1.6
06/01/ 16	5.8	1	3	41.1	-	-98	-61 to -87	5	1.8/1.53
15/09/ 16	2.4	2	4	51.3	-	-98	-53 to -85	5	2/5
23/09/ 16	2.4	2	4	52	-	-98	-51 to -82	5	3/5
26/09/ 16	2.4	4	5	45	-	-96	-61 to -83	5	3/5



Fig. 8. Field trials summary (6 trials, 14 fishing vessels).

actively using voice and video communication services.

The project is currently in the process of deploying the communication solution to a wider range of fishing vessels, aiming to scale up its utilisation. Encouraging fishermen to adopt this system for safety and business improvement has proven to be a challenging but rewarding endeavour. The research team recognised from the outset that the success of this technological solution hinges upon ongoing community engagement throughout the design, development, and operation phases. Therefore, they committed themselves to fostering awareness at the community level. However, interactions with community members revealed several challenges hindering the project's goals. These included a lack of awareness about the benefits of digital services, limited digital access, and difficulties in forming habits of continuous usage. Additionally, the project encountered resistance due to a lack of social influence within the community to adopt new technologies. In general, rural communities often exhibit a lack of awareness and motivation to use digital solutions.

Designing a social change model for transforming this rural fishing community into a smart, digitally enabled one proved to be a crucial, yet highly challenging, phase of the project. Factors such as broad stakeholder involvement, community awareness-raising initiatives, and aligning the solution with the community's specific characteristics laid the foundation for the development of the sustainable social change model proposed in this research.

#### 4.4. Sustainable social change outcome

The adoption of the proposed digital ecosystem relies on the continuous interaction of stakeholders with the technological solutions and systems developed for them. According to Dashkevych et al. (2023) [112], smart technologies are perceived as tools to achieve sustainability and improve the quality of life. Sustainability can be achieved when the proposed technology empowers the community by bridging the gap in digital capacity and digital outcomes. Community empowerment needs a holistic staged approach emphasising the digital capacity building of all the stakeholders involved in the community transformation. To achieve this goal, the study proposes a sustainable social change model for the adoption of the ODE for long-term community transformation through digital empowerment. Social change in this study refers to the transformation of the target society in terms of

their conventional communication and business practices [113]. The digital outcome of the community is dependent on their utilisation of digital services for emergency communication, enhancing business opportunities, skill development, digital capacity, increased motivational access, and increased awareness. To achieve the community's digital outcome, a social change model was introduced.

The detailed analysis of the community helped to structure the social change and the ICT integration model envisioned to digitally empower the community which is shown in Fig. 9. The community stakeholders are at the core of the model, who had to be persuaded through various levels of engagement to appreciate the numerous opportunities they would benefit from the solution. The model has three major sections, and the outcomes of each are as follows.

- Digital access to various services over the Internet benefits the community whenever they need them. For instance, Internet access at sea provided them with real-time communication opportunities during their occupational voyages, such as VoIP and instant messaging. It also provided different services such as potential fishing zone identification (using satellite images), remote health monitoring services, weather forecasting, sea-state information, disaster alerts, localisation and tracking of fishing vessels, etc.
- The collective digital outcome achieved through various methods aimed to enhance user motivation and capability. Short-term outcomes observed in the community included their willingness to cooperate with the research team in installing the equipment on their



# DIGITAL EMPOWERMENT



fishing vessels, as well as their participation in training sessions and testing the equipment during voyages. However, initially, the team noticed that fishermen would turn off the equipment when researchers were not present. One reason for this was their concern about power consumption of the equipment. They also raised questions such as, "Will this system record our voices or listen to us if turned on?" indicating a need for assurance regarding privacy while using the system. In the short term, a lack of experience with digital systems and services, along with a lack of trust in the proposed solution, were evident, resulting in minimal confidence among individual fishermen. Over the medium term, spanning eighteen months of usage, there has been a significant increase in their utilisation. Fishermen have begun accessing instant messaging services and providing feedback for improvements, indicating their growing acceptance of the solution and recognition of its usefulness. Their feedback and usage patterns demonstrate trust in the system and an understanding of its benefits. Based on the received feedback, the research team updated the user interface, adding features such as auto antenna rotation, resulting in increased usage. Users have expressed a positive perception of the system's utility, leading to more frequent use. Moreover, they have begun sharing their experiences, both positive and negative, with other fishermen and vessel owners, indicating development of a sense of ownership among them. With over 1000 h logged across 41 fishing trips, there is a clear trend of increasing digital utilisation and motivation to engage with the system while at sea. This transition from initial hesitation to embracing the solution and recognising its value is evident compared to the initial phase. The long-term outcome is yet to be studied and the research team is planning to perform it soon.

• Digital capacity enhancement within the fishing community proved to be a gradual process. While the research team didn't observe immediate and significant improvements in digital skills among fishermen and other stakeholders, their commitment and perseverance paid off over time. Frequent, well-structured training sessions were instrumental in developing essential digital skills and fostering a perception of ease of use among stakeholders. Active participation in these sessions not only enhanced their ability to use the system but also facilitated knowledge acquisition. This ultimately boosted collective and individual motivation, leading to a greater appreciation of the solution's usefulness. The positive impact of the project was evident at multiple levels within the community. At the individual level, training and education sessions enhanced user perception of the system. Users demonstrated noticeable improvement in skill development, motivation to use the solution, and overall receptiveness to the technology. Over time, positive beliefs and emotions surrounding system usage also emerged. Interpersonal communication between fishermen and vessel owners improved significantly. This enhanced support network fostered greater system adoption. At the community level, technology awareness rose, and community leaders and family members expressed strong support for the communication system. They recognised the value it provided in facilitating communication between fishermen at sea and their families back home. The involvement of family members, peers, and community leaders leveraged social influence to further increase motivation among fishermen. Community awareness sessions played a vital role in

educating stakeholders about the communication platform's necessity and its potential to enhance both safety and livelihoods. These sessions resonated particularly well with women and children, who were more enthusiastic about adopting the solution than other community members. For them, the ability to stay connected with loved ones at sea provided significant peace of mind. Similarly, awareness sessions conducted in schools fostered technology acceptance among the younger generation. Looking towards the long term, the project envisions fostering enhanced leadership skills within the fishing community. This will be achieved by providing improved access to information, empowering them to make informed decisions. Ultimately, this is expected to strengthen support systems for fishermen during emergencies. The following quotes from fishermen's family members exemplify their strong support:

- " ... It is good to have a device to alert the fishermen if their boat is anchored in the route of large ships to prevent disasters from collisions ... (A family member)"
- "... We would like to be able to talk with our family members (at the sea) in case of any emergencies at home ... (A family member)"

Table 4 showcases the current state of digital empowerment within the community, as measured by the specific indicators during the research. As the table illustrates, most envisioned digital empowerment metrics have been partially achieved, with some reaching full realisation. For instance, access to digital services at sea has been fully achieved, while motivation and consistent system access are yet to reach their full potential. Indicators like confidence in using the solution, trust in its effectiveness, increased online time at sea, and improved digital skills all point towards an enhanced digital capability among fishermen. Additionally, their use of instant messaging services, their willingness to suggest solutions to others, and internet usage at sea demonstrate progress in community-wide technology adoption. Overall, these positive indicators suggest that the digital empowerment of the fishing community is on a positive trajectory.

The OceanNet digital ecosystem (ODE) addresses various categories of the digital divide, as illustrated in Fig. 10. Aligned with several United Nations Sustainable Development Goals (SDGs), the ODE contributes to SDG 9 by embracing innovative solutions and fostering resilient infrastructure. Additionally, it supports SDG 8 by promoting sustainable economic growth, SDG 2 by advancing food security and market information, SDG 3 by promoting well-being and safety, SDG 4 by facilitating lifelong learning and specific skill sets t bridge the digital divide, SDG 5 by empowering women in the community through managing the smart warehouses, and SDG 14 by providing the means for sustainable use of the marine ecosystem [114].

The ODE ecosystem helped to address the lack of digital access that existed in the study fishing community. The lack of digital capacity of the selected fishermen and vessel captains was enhanced through digital skill training, frequent communication and collaboration with them during the co-design and by focusing on the system's ease of use by upgrading the system based on the feedback provided by the users. This

#### Table 4

Progress of the envisioned digital empowerment of the community.

Indicator	Digital Empowerment Index	Status
Access to digital services	Digital Access	Achieved
Motivation and attitude	Digital Capacity	Partially
improvement		achieved
Confidence in using the solution	Digital Capacity	Achieved
Improve in technology anxiety	Digital Capacity	Partially
		achieved
Trust in the proposed solution	Digital Capacity, Outcome	Partially
		achieved
Frequent usage of system	Digital Access, Capacity,	Partially
	Outcome	achieved
User engagement	Digital Capacity	Achieved
Increased online time	Digital Access, Outcome	Achieved
Improved digital skills	Digital Capacity	Partially
		achieved
Internet usage beyond 15 km	Digital Access, Capacity,	achieved
	Outcome	
Feedback provision for system improvement	Digital Outcome	Achieved
Use of instant messaging service	Digital Outcome	Achieved
Suggestion of the solution to	Digital Outcome	Partially
others	-	achieved

enhanced overall digital resource awareness and established new digital practices. This enhanced the overall digital outcome of the community by providing options (see. Fig 10) for increased safety, profit management, and increased resilience by enabling the adoption of the technology by the users.

# 5. Discussion

The first research question focused on building a digital ecosystem to achieve real-time disaster risk reduction and enhance resilience to ocean emergencies and this underscored the necessity for a communication system at the ocean, while the co-design process further elucidated the requirement for a digital ecosystem to facilitate real-time disaster risk reduction and bolster resilience to ocean emergencies. The ecosystem was tailor-made for the community, and their involvement throughout the design, development, testing, and deployment phases underscored the necessity for a social change model to drive the adoption and usage of the proposed solution. This endeavour also addressed the second research question: How can the fishing community be empowered to leverage the developed technologies for improved safety and livelihood? Addressing the second research question required focusing on integrating technologies such as IoT devices integrated with a Long Range Wi-Fi communication for enabling V2I and V2I networking, inclusive mobile applications and blockchain to enable the community to access the Internet in the ocean for digital service access and build business opportunities and livelihood resilience. Similar to the process for addressing the first research question, strategies such as co-design and active engagement strategies strengthen the target community's digital skills and enabled them to utilise the OceanNet Digital Ecosystem (ODE) during their voyages.

#### 6. Limitations

This research, while demonstrating valuable contributions, has limitations due to its scope. The implemented solution was a trial version deployed in a single fishing village. This approach restricts the generalisability of findings, as specific challenges within that community may not be universal. One such challenge is the fluid nature of crews. Fishermen and captains often switch between vessels, necessitating repeated training for new crew members on equipped boats. Similarly, vessel ownership can change hands, and new owners may not be as cooperative as the initial participants. Furthermore, the widespread use of wireless technology hinges on obtaining special government permissions. This highlights the need for continued support from government authorities.

Scaling the project to encompass multiple villages across coastal

regions requires tailoring the approach. While the core solution is adaptable to most Indian fishing communities, strategies for stakeholder participation, capacity building, trust building, and training intervals may need adjustments based on specific community characteristics. A more in-depth study of diverse coastal communities is necessary to refine the model and ensure it meets the varied needs of different villages.

It's important to acknowledge that the proposed technology offers support, but it's not a cure-all. For example, the system cannot provide information on seasonal fish availability. While it empowers fishermen with better decision-making tools and easier communication, their safety ultimately relies on their attentiveness to warnings (border crossings, cyclones) and their ability to take appropriate action. The Internet access for digital services on the ocean also presents potential downsides. Similar to trends observed among younger generations, excessive use of social media and entertainment services could distract from learning about the ocean and their profession, potentially jeopardizing safety during fishing operations. The e-commerce platform within the solution, while offering wider customer reach, might disrupt traditional business practices. This could force some individuals, like intermediaries, to adapt their business models.

#### 7. Conclusion

Coastal fishing communities face a multitude of challenges, including natural disasters, emergencies at sea, and a significant digital divide. This lack of digital access and capacity acts as a major barrier, preventing them from fully utilising digital resources that could improve their safety and livelihoods. The consequences of natural disasters are often exacerbated by this limited digital access. This research directly addresses these issues by introducing the OceanNet Digital Ecosystem (ODE). ODE is a comprehensive suite of technological interventions designed to improve digital access, capacity, and overall well-being within the coastal fishing community. The solution provides internet connectivity and access to essential digital services through its innovative heterogeneous, multi-layered communication system. Additionally, a blockchain-based e-commerce platform offers new business opportunities, fostering greater livelihood resilience for the fishermen.

To ensure technology acceptance and maximize impact, all these technological solutions are integrated with a unique and sustainable social change model. This approach not only enhances disaster resilience but also improves the overall quality of life for the community. Codesigning the solution alongside the community was crucial. This collaborative process facilitated a deep understanding of their specific needs and ensured successful diffusion of the technology. It also allowed for the involvement of various stakeholders, both at the micro and



# OceanNet Digital Ecosystem (ODE): Bridging the Digital Divide

Fig. 10. Digital divide categories addressed by OceanNet digital ecosystem.

macro levels, who are vital for successful adoption within the community.

Frequent interactions with users played a vital role in shaping the overall system design and development. These interactions highlighted the importance of social cohesion in the project's success. Engaging with the community also facilitated capacity-building initiatives. This included school visits and educational sessions aimed at empowering the younger generation and fostering awareness about the solution. Regularly collecting feedback and suggestions from fishing vessel captains and fishers proved beneficial. This built trust and empowered them to provide clear and specific requirements and suggestions, ultimately aiding the research team throughout the deployment and training process. By employing various methodologies, this study has identified key factors and developed intervention strategies that contribute to successful technology transfer in such contexts.

Future endeavours aim to expand the mobile application's functionalities by incorporating additional modules that address various social and business needs within the fishing community. This will enhance the app's overall value proposition. Scaling the project is a key priority. Efforts are underway to deploy the communication system across over a hundred fishing vessels. This large-scale implementation will serve to test the system's robustness and reliability in real-world operational conditions. The team is actively working on the e-commerce platform, with the blockchain structure and its framework design already finalised. The next step involves identifying suitable entities to act as blockchain ledger holders, paving the way for full implementation. In a significant collaboration, the research team is partnering with the Indian National Centre for Ocean Information Services (INCOIS) to implement the Tsunami- Ready Recognition program [115, 116]. This program will equip fishing villages with various mitigation measures, ultimately reducing their vulnerability and increasing their resilience to natural hazards, particularly tsunamis.

#### Acknowledgements

The authors would like to express their immense gratitude to Sri. Mata Amritanandamayi Devi (AMMA), Chancellor of Amrita Vishwa Vidyapeetham, for her valuable ideas and implementation suggestions that contributed to this research. This research was partially funded by a grant titled "Mobile Infrastructure for Coastal Region Offshore Communications & Networks" (ITRA/15(65)/Mobile/MicroNet/01) from the Information Technology Research Agency (ITRA), Department of Electronics and Information Technology (DeitY), Government of India, and Amrita Vishwa Vidyapeetham.

#### CRediT authorship contribution statement

Sruthy Anand: Writing – review & editing, Writing – original draft, Software, Methodology, Investigation, Conceptualization. Mojtaba Enayati: Writing – review & editing, Software, Methodology. Dhanesh Raj: Writing – original draft, Validation, Software, Data curation. Alberto Montresor: Writing – review & editing, Supervision. Maneesha Vinodini Ramesh: Writing – review & editing, Writing – original draft, Validation, Supervision, Project administration, Funding acquisition, Conceptualization.

#### Data availability

The data that has been used is confidential.

#### References

- Ranadhir Mukhopadhyay, S.M. Karisiddaiah, The Indian Coastline: Processes and Landforms, Springer, Netherlands, 2014, pp. 91–101, https://doi.org/10.1007/ 978-94-017-8029-2\_8.
- [2] Animal Husbandry & Dairying Government of India Fisheries Statistics Division Department of Fisheries Ministry of Fisheries. *Handbook on Fisheries Statistics* 2020, Ministry of Fisheries; Government of India, 2020.

- [3] Rekha Nair, Status of Marine Fisheries of kerala. Marine Fisheries Information Service; Technical and Extension Series (226), Marine Fisheries Information Service; Technical and Extension Series, 2016, pp. 22–26.
- [4] E. Carolyn, Sachs. Women Working in the Environment: Resourceful Natures, Routledge, 2017.
- [5] Richard B. Pollnac, Iris Monnereau, John J. Poggie, Victor Ruiz, Azure D. Westwood, 15 stress and the occupation of fishing, Handbook of stress in the occupations 309 (2011).
- [6] Lijing Cheng, Karina von Schuckmann, John P. Abraham, Kevin E. Trenberth, Michael E. Mann, Laure Zanna, Matthew H. England, Jan D. Zika, John T. Fasullo, Yongqiang Yu, et al., Past and future ocean warming, Nat. Rev. Earth Environ. 3 (11) (2022) 776–794.
- [7] Tribune News Service, Over 1,00,000 fishing-related deaths occur annually: report, URL, https://www.tribuneindia.com/news/world/over-1-00-000-fishin g-related-deaths-occur-annually-report-447284, 2023.
- [8] S.N. Rao, M.V. Ramesh, V. Rangan, Mobile infrastructure for coastal region offshore communications and networks, in: 2016 IEEE Global Humanitarian Technology Conference (GHTC), Oct 2016, pp. 99–104, https://doi.org/10.1109/ GHTC.2016.7857266.
- [9] Susan L. Cutter, Lindsey Barnes, Melissa Berry, Christopher Burton, Elijah Evans, Eric Tate, Jennifer Webb, A place-based model for understanding community resilience to natural disasters, Global Environ. Change 18 (4) (2008) 598–606.
- [10] Aijun Guo, Yao Wei, Fanglei Zhong, Penglong Wang, How do climate change perception and value cognition affect farmers' sustainable livelihood capacity? an analysis based on an improved dfid sustainable livelihood framework, Sustain. Prod. Consum. 33 (2022) 636–650.
- [11] Mojtaba Enayati, Sudha Arlikatti, Maneesha Vinodini Ramesh, A qualitative analysis of rural fishermen: potential for blockchain-enabled framework for livelihood sustainability, Heliyon 10 (2) (2024) e24358.
- [12] Dineshkumar Singh, Mkrishi fisheries: sea in your hand, URL, https://impakter.co m/mkrishi-fisheries-sea-hand/, May 2019.
- [13] Tides, wind, waves, solunar, marine +, URL, https://nautide.com/, 2023.
- [14] Esso-INCOIS-Indian National Centre for Ocean Information Services, Search and rescue aid tool to save life and property at sea, URL, https://incois.gov.in/portal/ osf/sarat.jsp, 2022.
- [15] Esso-INCOIS-Indian National Centre for Ocean Information Services, "sagar vani"

   an integrated information dissemination system, URL, https://incois.gov.in/port al/datainfo/ids.jsp, 2022.
- [16] Live wind map & weather forecast, URL, https://windy.app/, 2023.
- [17] J. Calderwood, Smartphone application use in commercial wild capture fisheries, Reviews in Fish Biology and Fisheries 32 (4) (2022) 1063–1083.
- [18] A. Mahesh, Ganesh Bhat, Digital payment service in India a case study of unified payment interface, International Journal of Case Studies in Business, IT, and Education (2021) 256–265, https://doi.org/10.47992/IJCSBE.2581.6942.0114.
- [19] Lisha Ye, Huiqin Yang, From digital divide to social inclusion: a tale of mobile platform empowerment in rural areas. Sustainability 12 (6) (2020) 2424.
- [20] Balwant Singh Mehta, Impact of mobile phone on livelihood of rural people, J. Rural Dev. (2016) 483–505.
- [21] Anique Scheerder, Alexander Van Deursen, Jan Van Dijk, Determinants of internet skills, uses and outcomes. a systematic review of the second-and thirdlevel digital divide, Telematics Inf. 34 (8) (2017) 1607–1624.
- [22] Marius Hagen, Herman Melbye, The Emergence of Digital Ecosystems: an Exploratory Case Study, Master's thesis, Norwegian School of Economics, 2020.
- [23] Mariana Petrova, Petya Popova, Veselin Popov, Krasimir Shishmanov, Kremena Marinova, Digital ecosystem: nature, types and opportunities for value creation, in: International Scientific Conference on Innovations in Digital Economy, Springer, 2021, pp. 71–85.
- [24] Timothy R. Frankenberger, M. Katherine McCaston, The Household Livelihood Security Concept. Food Nutrition and Agriculture, 1998, pp. 30–35.
- [25] Brian Walker, Crawford S. Holling, Stephen R. Carpenter, Ann Kinzig, Resilience, adaptability and transformability in social–ecological systems, Ecol. Soc. 9 (2) (2004).
- [26] Gitz Vincent, Alexandre Meybeck, et al., Risks, vulnerabilities and resilience in a context of climate change, Building resilience for adaptation to climate change in the agriculture sector 23 (2012) 19.
- [27] Robert Chambers, Gordon Conway, Sustainable Rural Livelihoods: Practical Concepts for the 21st Century, Institute of Development, Studies (UK), 1992.
- [28] Wirawat Chaya, Shabbir H. Gheewala, Sustainable livelihood outcomes, causal mechanisms and indicators self-determined by Thai farmers producing bioethanol feedstocks, Sustain. Prod. Consum. 29 (2022) 447–466.
- [29] Nina Aarras, Mia Rönkä, Matti Kamppinen, Harri Tolvanen, Petteri Vihervaara, Environmental technology and regional sustainability – the role of life-based design, Technol. Soc. 36 (2014) 52–59, https://doi.org/10.1016/j. techsoc.2013.12.003.
- [30] Fred D. Davis, User acceptance of information technology: system characteristics, user perceptions and behavioral impacts, Int. J. Man Mach. Stud. 38 (3) (1993) 475–487, https://doi.org/10.1006/imms.1993.1022.
- [31] Census of India. https://censusindia.gov.in/census.website/data/population-finder, 2011.
- [32] Marc Steen, Menno Manschot, Nicole De Koning, Benefits of co-design in service design projects, Int. J. Des. 5 (2) (2011).
- [33] Marc Steen, Co-design as a process of joint inquiry and imagination, Des. Issues 29 (2) (2013) 16–28.
- [34] Tony McAleavy, Perceiving the effects of scale on command and control: a conceptual metaphor approach, Journal of Emergency Management (Weston, Mass.) 18 (2) (2020) 91–104.

Technology in Society 79 (2024) 102686

- [35] Tony McAleavy, Martin Rhisiart, Harnessing the power of metaphor: Uncovering a hidden language of interoperability within the natural speech of emergency managers, Int. J. Emerg, Manag. 15 (1) (2019) 1–25.
- [36] Ben Purvis, Yong Mao, Darren Robinson, Three pillars of sustainability: in search of conceptual origins, Sustain. Sci. 14 (3) (2019) 681–695.
- [37] Lorenz Graf-Vlachy, Katharina Buhtz, Andreas König, Social influence in technology adoption: taking stock and moving forward, Management Review Quarterly 68 (2018) 37–76.
- [38] Ibrahim Arpaci, Kasim Karatas, Ismail Kusci, Mostafa Al-Emran, Understanding the social sustainability of the metaverse by integrating utaut2 and big five personality traits: a hybrid sem-ann approach, Technol. Soc. 71 (2022) 102120.
- [39] Sophie Lythreatis, Sanjay Kumar Singh, Abdul-Nasser El-Kassar, The digital divide: a review and future research agenda, Technol. Forecast. Soc. Change 175 (2022) 121359.
- [40] Ling Zhao, Cuicui Cao, Yuni Li, Li Yuan, Determinants of the digital outcome divide in e-learning between rural and urban students: empirical evidence from the covid-19 pandemic based on capital theory, Comput. Hum. Behav. 130 (2022) 107177.
- [41] Jonathan Morris, Wyn Morris, Robert Bowen, Implications of the digital divide on rural sme resilience, J. Rural Stud. 89 (2022) 369–377.
- [42] H.S. Jennath, Nandesh Nair, Sethuraman Rao, Dhanesh Raj, Sujatha Narayanan, Dilip Krishnaswamy, A resilient self-organizing offshore communication network for fishermen, in: *Proceedings of the 18th International Conference on Distributed Computing and Networking*, ICDCN '17, Association for Computing Machinery, New York, NY, USA, 2017, https://doi.org/10.1145/3007748.3018283.
- [43] Vrinda N. Menon, Anagha Suresh, Analyzing digital overlay technique over marine voice channel for disaster dissemination, in: 2014 Eleventh International Conference on Wireless and Optical Communications Networks (WOCN), 2014, pp. 1–5, https://doi.org/10.1109/WOCN.2014.6923090.
- [44] K. Rakesh, S. Sunanda, The development of a visualization framework for fishing vessels at sea, in: Fifth International Conference on Computing, Communications and Networking Technologies (ICCCNT), 2014, pp. 1–5, https://doi.org/ 10.1109/ICCCNT.2014.6963077.
- [45] Dhanesh Raj, Maneesha Vinodini Ramesh, Sethuraman N. Rao, Link quality measurements for marine vehicular to infrastructure network (v2i), in: 2020 International Conference on COMmunication Systems & NETworkS (COMSNETS), 2020, pp. 578–581, https://doi.org/10.1109/ COMSNETS48256.2020.9027379.
- [46] Mojtaba Enayati, S. Sree Lekshmi, Bhanu Prakash Poluparthi, Lakshmi Rajagopal, Seshaiah Ponnekanti, Integrating blockchain, open ran and iot platforms into rural community networks, in: Proceedings of the International Conference on Sustainable Development in Technology for 4th Industrial Revolution, 2021.
- [47] Tuheen Ahmmed, Afsoon Alidadi, Zichao Zhang, Aizaz U. Chaudhry, Halim Yanikomeroglu, The digital divide in Canada and the role of leo satellites in bridging the gap, IEEE Commun. Mag. 60 (6) (2022) 24–30.
- [48] Muhammad Waseem Akhtar, Nasir Saeed, Uavs-enabled maritime communications: uavs-enabled maritime communications: opportu- nities and challenges, IEEE Systems, Man, and Cybernetics Magazine 9 (3) (2023) 2–8, https://doi.org/10.1109/MSMC.2022.3231415.
- [49] Mostafa Al-Emran, Charla Griffy-Brown, The role of technology adoption in sustainable development: overview, opportunities, challenges, and future research agendas, Technol. Soc. 102240 (2023).
- [50] Ester Van Laar, Alexander JAM. Van Deursen, Jan Agm Van Dijk, Jos de Haan, Determinants of 21st-century skills and 21st-century digital skills for workers: a systematic literature review, Sage Open 10 (1) (2020) 2158244019900176.
- [51] Shahla Ghobadi, Zahra Ghobadi, How access gaps interact and shape digital divide: a cognitive investigation, Behav. Inf. Technol. 34 (4) (2015) 330–340.
- [52] Jan AGM Van Dijk, Digital divide: impact of access, in: The International Encyclopedia of Media Effects, 2017, pp. 1–11.
- [53] Ahmed Soomro Kamal, Ugur Kale, Reagan Curtis, Mete Akcaoglu, Malayna Bernstein, Digital divide among higher education faculty, International Journal of Educational Technology in Higher Education 17 (2020) 1–16.
- [54] Deepak Suresh Varma, Nandanan Krishna, P.C. Vishakh Raja, B. Soundharajan, Mireia López Pérez, K.A. Sidharth, Maneesha Vinodini Ramesh, Participatory design approach to address water crisis in the village of karkatta, Jharkhand, India, Technol. Forecast. Soc. Change 172 (2021) 121002, https://doi.org/ 10.1016/j.techfore.2021.121002.
- [55] Vineeth Ajith, A.S. Reshma, Renjith Mohan, Maneesha Vinodini Ramesh, Empowering communities in addressing drinking water challenges using a systematic, participatory and adaptive approach and sustainable ppp model, Technol. Forecast. Soc. Change 185 (2022) 121970, https://doi.org/10.1016/j. techfore.2022.121970.
- [56] Sonia Livingstone, Giovanna Mascheroni, Mariya Stoilova, The outcomes of gaining digital skills for young people's lives and wellbeing: a systematic evidence review, New Media Soc. 25 (5) (2023) 1176–1202.
- [57] Shantanu Sharma, Devika Mehra, Faiyaz Akhtar, Sunil Mehra, Evaluation of a community-based intervention for health and economic empowerment of marginalized women in India, BMC Publ. Health 20 (1) (2020) 1–16.
- [58] Maria E. Mondejar, Ram Avtar, Heyker Lellani Baños Diaz, Rama Kant Dubey, Jesús Esteban, Abigail Gómez-Morales, Brett Hallam, Nsilulu Tresor Mbungu, Chukwuebuka Christopher Okolo, Kumar Arun Prasad, et al., Digitalization to achieve sustainable development goals: steps towards a smart green planet, Sci. Total Environ. 794 (2021) 148539.
- [59] Jeong Seop Hwang, Jae Jeung Rho, Yoon Min Hwang, Influence of cognitive and social change factors on e-vehicle switching intention: evidence from korea, Technol. Soc. 102286 (2023).

- [60] Marija Cubric, Drivers, barriers and social considerations for ai adoption in business and management: a tertiary study, Technol. Soc. 62 (2020) 101257.
- [61] Manindra Rajak, Krishnendu Shaw, An extension of technology acceptance model for mhealth user adoption, Technol. Soc. 67 (2021) 101800.
- [62] Tai-Kuei Yu, Mei-Lan Lin, Ying-Kai Liao, Understanding factors influencing information communication technology adoption behavior: the moderators of information literacy and digital skills, Comput. Hum. Behav. 71 (2017) 196–208, https://doi.org/10.1016/j.chb.2017.02.005. ISSN 0747-5632, https://www.sci encedirect.com/science/article/pii/S074756321730078X.
- [63] Flor Avelino, Theories of power and social change. power contestations and their implications for research on social change and innovation, Journal of Political Power 14 (3) (2021) 425–448.
- [64] Sandra A. Vannoy, Prashant Palvia, The social influence model of technology adoption, Commun. ACM 53 (6) (2010) 149–153.
- [65] Ioana Iancu, Bogdan Iancu, Designing mobile technology for elderly. a theoretical overview, Technol. Forecast. Soc. Change 155 (2020) 119977.
- [66] Zereh Lalji, Judith Good, Designing new technologies for illiterate populations: a study in mobile phone interface design, Interact. Comput. 20 (6) (2008) 574–586.
- [67] Aygül Dönmez-Turan, Merve Kır, User anxiety as an external variable of technology acceptance model: a meta-analytic study, Proc. Comput. Sci. 158 (2019) 715–724, https://doi.org/10.1016/j.procs.2019.09.107.
- [68] Liaquat Hossain, Anjali de Silva, Exploring user acceptance of technology using social networks, J. High Technol. Manag. Res. 20 (1) (2009) 1–18, https://doi. org/10.1016/j.hitech.2009.02.005.
- [69] Andrew Dillon, Michael G. Morris, User acceptance of new information technology: theories and models (1996). URL.
- [70] Fred D. Davis, Richard P. Bagozzi, Paul R. Warshaw, User acceptance of computer technology: a comparison of two theoretical models, Manag. Sci. 35 (8) (1989) 982–1003, https://doi.org/10.1287/mnsc.35.8.982.
- [71] Taherdoost Hamed, A review of technology acceptance and adoption models and theories, Procedia Manuf. 22 (2018) 960–967, https://doi.org/10.1016/j. promfg.2018.03.137.11thInternational. Conference Interdisciplinarity in Engineering, INTER-ENG 2017, 5-6 October 2017, Tirgu Mures, Romania.
- [72] Joyce D. Jackson, Mun Y. Yi, Jae S. Park, An empirical test of three mediation models for the relationship between personal innovativeness and user acceptance of technology, Inf. Manag. 50 (4) (2013) 154–161, https://doi.org/10.1016/j. im.2013.02.006.
- [73] Janine Chung, Felix B. Tan, Antecedents of perceived playfulness: an exploratory study on user acceptance of general information- searching websites, Inf. Manag. 41 (7) (2004) 869–881, https://doi.org/10.1016/j.im.2003.08.016.
- [74] Ronnie Cheung, Doug Vogel, Predicting user acceptance of collaborative technologies: an extension of the technology acceptance model for e-learning, Comput. Educ. 63 (2013) 160–175, https://doi.org/10.1016/j. compedu.2012.12.003.
- [75] Astrid Albrecht, Understanding the issues behind user acceptance, Biom. Technol. Today 9 (1) (2001) 7–8, https://doi.org/10.1016/S0969-4765(01)00124-2.
- [76] Trino Jusuf Habibie, Rahmat Yasirandi, Dita Oktaria, The analysis of pangandaran fisherman's actual usage level of gps based on tam model, Proc. Comput. Sci. 197 (2022) 34–41, https://doi.org/10.1016/j.procs.2021.12.115. SixthInformationSystemsInternationalConference (ISICO 2021).
- [77] Heetae Yang, Jieun Yu, Hangjung Zo, Munkee Choi, User acceptance of wearable devices: an extended perspective of perceived value, Telematics Inf. 33 (2) (2016) 256–269, https://doi.org/10.1016/j.tele.2015.08.007.
  [78] Yunzhou Li, Ming Sun, Yiping Ren, Yong Chen, Fisher behavior matters:
- [78] Yunzhou Li, Ming Sun, Yiping Ren, Yong Chen, Fisher behavior matters: harnessing spatio-temporal fishing effort information to support China's fisheries management, Ocean Coast Manag. 210 (2021) 105665, https://doi.org/10.1016/ j.ocecoaman.2021.105665.
- [79] J. Hartwick, H. Barki, Communication as a dimension of user participation, IEEE Trans. Prof. Commun. 44 (1) (March 2001) 21–36, https://doi.org/10.1109/ 47.911130.
- [80] Fred D. Davis, User acceptance of information technology: system characteristics, user perceptions and behavioral impacts, Int. J. Man Mach. Stud. 38 (3) (1993) 475–487, https://doi.org/10.1006/imms.1993.1022.
- [81] Sujeet Kumar Sharma, Ali H. Al-Badi, Srikrishna Madhumohan Govindaluri, Mohammed H. Al-Kharusi, Predicting motivators of cloud computing adoption: a developing country perspective, Comput. Hum. Behav. 62 (2016) 61–69, https:// doi.org/10.1016/j.chb.2016.03.073.
- [82] Tino Fenech, Using perceived ease of use and perceived usefulness to predict acceptance of the world wide web, Comput. Netw. ISDN Syst. 30 (1) (1998) 629–630, https://doi.org/10.1016/S0169-7552(98)00028-2. Proceedings of the Seventh International World Wide Web Conference.
- [83] S.S. Sreeraj, A. Unnikrishnan, K. Vishnu, Noah E. Kennith, Sruthy Anand, Maneesha Vinodini Ramesh, Empowerment of women self help groups: human centered design of a participatory iot solution, in: 2020 IEEE Global Humanitarian Technology Conference (GHTC), IEEE, 2020, pp. 1–8.
- [84] Sarah Anrijs, Ilse Mariën, Lieven De Marez, Koen Ponnet, Excluded from essential internet services: examining associations between digital exclusion, socioeconomic resources and internet resources, Technol. Soc. 73 (2023) 102211.
- [85] Carmen Bruno, Marita Canina, Creativity 4.0. empowering creative process for digitally enhanced people, Des. J. 22 (sup1) (2019) 2119–2131.
- [86] Cecilia Perri, Carlo Giglio, Vincenzo Corvello, Smart users for smart technologies: investigating the intention to adopt smart energy consumption behaviors, Technol. Forecast. Soc. Change 155 (2020) 119991.
- [87] Syeda Ayesha Kamal, Muhammad Shafiq, Priyanka Kakria, Investigating acceptance of telemedicine services through an extended technology acceptance model (tam), Technol. Soc. 60 (2020) 101212.

Technology in Society 79 (2024) 102686

- [88] Susmita Chatterjee, Sangita Dutta Gupta, Parijat Upadhyay, Technology adoption and entrepreneurial orientation for rural women: evidence from India, Technol. Forecast. Soc. Change 160 (2020) 120236.
- [89] Roxane De la Sablonniere, Toward a psychology of social change: a typology of social change, Front. Psychol. 8 (2017) 226192.
- [90] Charles Tilly, Social Movements, 1768–2004, Routledge, 2019, https://doi.org/ 10.4324/9781315632063. ISBN 9781315632063.
- [91] Neil J. Smelser, Social transformations and social change, Int. Soc. Sci. J. 50 (156) (1998).
- [92] Jason T. Newsom, Ann McQueen, Karen S. Rook, Neal Krause, Emily C. Denning, A change for the worse: negative social exchanges are associated with an accelerated decline in self-rated health over time, J. Aging Health 34 (6–8) (2022) 984–995.
- [93] Theda Skocpol, States and Social Revolutions: A Comparative Analysis of France, Russia and China, Cambridge University Press, 1979.
- [94] C Devezas Tessaleno, Evolutionary theory of technological change: state-of-theart and new approaches, Technol. Forecast. Soc. Change 72 (9) (2005) 1137–1152.
- [95] Gisela Trommsdorff, Effects of social change on individual development: the role of social and personal factors and the timing of events, Negotiating adolescence in times of social change (2000) 58–68.
- [96] Jürgen Howaldt, Ralf Kopp, Michael Schwarz, Social Innovations as Drivers of Social Change—Exploring Tarde's Contribution to Social Innovation Theory Building, Palgrave Macmillan UK, 2015.
- [97] Diane E. Alington, Lillian E. Troll, Social change and equality: the roles of women and economics, in: Women in Midlife, Springer, 1984, pp. 181–202.
- [98] Wendy R. Williams, Kala J. Melchiori, Class action: using experiential learning to raise awareness of social class privilege, in: Deconstructing Privilege, Routledge, 2013, pp. 169–187.
- [99] Everett M. Rogers, Arvind Singhal, Margaret M. Quinlan, Diffusion of Innovations. Rogers, vol. 576, Simon and Schuster, NY, 2003, 2003.
- [100] David Baker, Social theory and the coming schooled society. Education in a New Society: Renewing the Sociology of Education, 2018, pp. 61–82.
- [101] Elaine Ho, Amelia Clarke, Ilona Dougherty, Youth-led social change: topics, engagement types, organizational types, strategies, and impacts, Futures 67 (2015) 52–62.
- [102] Sarah Schulman, Terms of engagement: aligning youth, adults, and organizations toward social change, J. Publ. Health Manag. Pract. 12 (2006) S26–S31.
- [103] Sruthy Anand, Dhanesh Raj, AM Abhishek Sai, Sethuraman N. Rao, Maneesha Vinodini Ramesh, Techno-social synergy for disaster resilience in coastal communities: a sustainable approach, in: 24th International Conference on Distributed Computing and Networking, 2023, pp. 366–371.
- [104] Elena Bessonova, Yulia Kelesh, Alexey Babichev, Shaping an effective ecosystem of the regional digital economy in the context of uneven digital development, in: Comprehensible Science: ICCS 2021, Springer, 2022, pp. 207–218.
- [105] D. Raj, M.V. Ramesh, S. Duttagupta, Delay tolerant routing protocol for heterogeneous marine vehicular mobile ad-hoc network, in: 2017 IEEE International Conference on Pervasive Computing and Communications Workshops (PerCom Workshops), March 2017, pp. 461–466, https://doi.org/ 10.1109/PERCOMW.2017.7917607.
- [106] Neena Joseph, Missing link between state and community: post-tsunami reconstruction and rehabilitation of alappad panchayat, Kerala, India. Strategic Disaster Risk Management in Asia, 2015, pp. 129–149.
- [107] Brian Walker, Crawford S. Holling, Stephen R. Carpenter, Ann Kinzig, Resilience, adaptability and transformability in social–ecological systems, Ecol. Soc. 9 (2) (2004).
- [108] Revin Naufal Alief, Muhammad Rasyid Redha Ansori, Ikechi Saviour Igboanusi, Jae Min Lee, Dong-Seong Kim, et al., Blockchain-based lightweight fish auction marketplace platform for traceability seafood supply chain. Proceedings of the Korean Institute of Communications and Information Sciences Conference, 2023, pp. 718–720.
- [109] Benjamin S. Thompson, Sascha Rust, Blocking blockchain: examining the social, cultural, and institutional factors causing innovation resistance to digital technology in seafood supply chains, Technol. Soc. 73 (2023) 102235.
- [110] Md Akhtaruzzaman Khan, Md Emran Hossain, Shahaab Ali, Imtiaz Khan Shrimpchain, A blockchain-based transparent and traceable framework to enhance the export potentiality of bangladeshi shrimp, Smart Agricultural Technology 2 (2022) 100041.
- [111] D. Reddy, V. Parthasarathy, S.N. Rao, Analysis of the effect of waves on the stability of tdma based marine long range wi-fi backhaul links, in: 2016 IEEE International Conference on Computational Intelligence and Computing Research (ICCIC), Dec 2016, pp. 1–8, https://doi.org/10.1109/ICCIC.2016.7919646.
- [112] Oleg Dashkevych, Boris A. Portnov, Human-centric, sustainability-driven approach to ranking smart cities worldwide, Technol. Soc. 102296 (2023).
- [113] Brady Wagoner, Séamus A Power. Social change, in: The Palgrave Encyclopedia of the Possible, Springer, 2023, pp. 1503–1508.
- [114] Department of Economic United Nations and Social Affairs Sustainable Development, Transforming our world: the 2030 agenda for sustainable development, URL, https://sdgs.un.org/2030agenda, 2015.
- [115] UNESCO, Tsunami ready recognition programme, Available at: http://itic. ioc-unesco.org/index.php?option=com\_content&view=category&id=22 34&Itemid=2758(2022/06/12, 2022.
- [116] Ready Tsunami. Mou with indian national centre for ocean information services (incois). Available at https://twitter.com/AMRITAedu/status/1535539139 143036928, June 2022.



Sruthy Anand is a PhD student at Amrita Vishwa Vidyapeetham and a Senior Research Associate at the Center for Wireless Networks and Applications in Amritapuri. She holds double master's degree, one in Computer Science and another in Wireless Networks and Applications. She started her professional career as a software developer and then moved to research and academics. Her areas of interest include dynamic vehicular networks and edge learning, disaster risk reduction, smart city, sustainable development, chatbots and designing software solutions for rural communities.



Mojtaba Enayati obtained his B.S. in computer hardware from Shomal University, Iran, followed by a Master of Computer Application from Savitribai Phule Pune University, India, in 2014. Currently, he is in the final year of his Ph.D. in Sustainable Development at Amrita Vishwa Vidyapeetham, India, with a primary focus on blockchain technology. His research endeavours revolve around leveraging blockchain technology to enhance sustainability and resilience among fisherfolk in rural Indian communities. He is particularly intrigued by the development of sustainable blockchain-based solutions aimed at fostering a more resilient digital ecosystem.



Dhanesh Raj is a PhD student at Amrita Vishwa Vidyapeetham and a Research Associate at the Center for Wireless Networks and Applications in Amritapuri. He is dedicated to developing a mobile infrastructure for offshore communications and networking in the coastal region to improve communication during natural disasters and hazards. His research focuses on Quality of Service (QoS) in marine 5G heterogeneous networks, private networks for 5G and mission-critical services, and community networks. He is a phase II award winner of 5G Hackathon 2021, conducted by DoT, India



Alberto Montresor received his Ph.D. in Computer Science from the University of Bologna, Italy in 2000. He joined the University of Trento in 2005 and became a Full Professor in 2020. Since 2022, he has been a member of the Academic Senate and serves as the "Delegate of the Rector "for Third Mission at the University of Trento. Montresor has published over 120 papers in the field of distributed systems, with a focus on large-scale decentralised systems. He was recognised as a Highly Cited Researcher in both 2020 and 2022 by Stanford University's Highly Cited Researchers Ranking.



Maneesha Vinodini Ramesh received the Ph.D. degree in computer science and engineering from Amrita Vishwa Vidyapeetham University in 2009. She currently serves Provost of Amrita Vishwa Vidyapeetham, holds multiple portfolios such as Strategic Initiatives, Research & Innovation, Interdisciplinary and AI + X. She is the Dean of International Programs and School for Sustainable Futures, and the Director & Professor at Amrita Center for Wireless Networks and Applications (AWNA). She was one of the Co-Principal Investigator of the European Commission-funded Wireless Sensor Networks with Self Organization Capabilities for Critical and Emergency Applications (WINSOC) Project, and principal investigator of eight internationally recognised projects funded by different

organizations from all over the world. She has more than 170 publications, including several journals and best paper awards. She was a recipient of the NABARD Award for Rural Innovation—Second Prize from the Honorable Finance Minister, Government of India, in 2012, for her research activities that benefited the rural community. She is a member of the Task Force for Formulation of National Landslide Risk Management Strategy, Government of India. She is an Editor of the Ad Hoc Networks Elsevier and CSI Transactions on ICT Springer. She has given invited talks at several eminent universities all over the world.